Neurosciences and Wireless Networks: The Potential of Brain-Type Communications and Their Applications

Renan Cipriano Moioli[®], Pedro H. J. Nardelli[®], Senior Member, IEEE,

Michael Taynnan Barros^(D), *Member, IEEE*, Walid Saad^(D), *Fellow, IEEE*, Amin Hekmatmanesh^(D),

Pedro E. Gória Silva¹⁰, Arthur Sousa de Sena¹⁰, Student Member, IEEE, Merim Dzaferagic¹⁰,

Harun Siljak[®], Senior Member, IEEE, Werner Van Leekwijck, Dick Carrillo Melgarejo[®], Member, IEEE, and Steven Latré, Senior Member, IEEE

Abstract—This paper presents the first comprehensive tutorial on a promising research field located at the frontier of two wellestablished domains, neurosciences and wireless communications, motivated by the ongoing efforts to define the Sixth Generation of Mobile Networks (6G). In particular, this tutorial first provides a novel integrative approach that bridges the gap between these two seemingly disparate fields. Then, we present the stateof-the-art and key challenges of these two topics. In particular, we propose a novel systematization that divides the contributions into two groups, one focused on what neurosciences will offer to future wireless technologies in terms of new applications and systems architecture (Neurosciences for Wireless Networks), and the other on how wireless communication theory and nextgeneration wireless systems can provide new ways to study the brain (Wireless Networks for Neurosciences). For the first group, we explain concretely how current scientific understanding of the brain would enable new applications within the context of a new type of service that we dub brain-type communications and that

Manuscript received October 9, 2020; revised April 28, 2021; accepted June 13, 2021. Date of publication June 23, 2021; date of current version August 23, 2021. This work was supported in part by the Academy of Finland through ee-IoT Project under Grant 319009, FIREMAN Consortium under Grant CHIST-ERA-17-BDSI-003/326270, and EnergyNet Research Fellowship under Grant 321265 and Grant 328869; in part by the EDUFI Visiting Researcher Grant; in part by the JAES Foundation in Finland through STREAM Project; in part by the Science Foundation Ireland (SFI) for the CONNECT Research Centre under Grant 13/RC/2077_P2; in part by the U.S. National Science Foundation under Grant CNS-1836802; and in part by the European Union's Horizon 2020 Research and Innovation Programme through the Marie Skłodowska-Curie Grant under Agreement 839553. (*Corresponding author: Pedro H. J. Nardelli.*)

Renan Cipriano Moioli is with the Department of EE, Federal University of Rio Grande do Norte, Natal 59064-720, Brazil.

Pedro H. J. Nardelli, Amin Hekmatmanesh, Arthur Sousa de Sena, and Dick Carrillo Melgarejo are with the School of Energy Systems, LUT University, 53850 Lappeenranta, Finland (e-mail: pedro.nardelli@lut.fi).

Michael Taynnan Barros is with the BioMediTech Institute, Tampere University, 33100 Tampere, Finland, and also with the School of Computer Science and Electronic Engineering, University of Essex, Colchester CO4 3SQ, U.K.

Walid Saad is with the Department of ECE, Virginia Tech, Blacksburg, VA 24061 USA.

Pedro E. Gória Silva is with the School of Energy Systems, LUT University, 53850 Lappeenranta, Finland, and also with the National Telecommunications Institute, Santa Rita do Sapucaí 37540-000, Brazil.

Merim Dzaferagic and Harun Siljak are with the CONNECT Centre, Trinity College, Dublin 2, D02 PN40 Ireland.

Werner Van Leekwijck and Steven Latré are with the imec, Antwerp University, 2000 Antwerpen, Belgium.

Digital Object Identifier 10.1109/COMST.2021.3090778

has more stringent requirements than human- and machine-type communication. In this regard, we expose the key requirements of brain-type communication services and discuss how future wireless networks can be equipped to deal with such services. Meanwhile, for the second group, we thoroughly explore modern communication systems paradigms, including Internet of Bio-Nano Things and wireless-integrated brain-machine interfaces, in addition to highlighting how complex systems tools can help bridging the upcoming advances of wireless technologies and applications of neurosciences. Brain-controlled vehicles are then presented as our case study to demonstrate for both groups the potential created by the convergence of neurosciences and wireless communications, probably in 6G. In summary, this tutorial is expected to provide a largely missing articulation between neurosciences and wireless communications while delineating concrete ways to move forward in such an interdisciplinary endeavor.

Index Terms—Wireless communications, neurosciences, braintype communications, brain-controlled vehicles, brain-machine interfaces, brain implants.

ACRONYMS

6G	Sixth Generation of Mobile Networks			
AI	Artificial Intelligence			
ANN	Artificial Neural Network			
AoI	Age of Information			
BCV	Brain-Controlled Vehicles			
BMI	Brain-Machine Interface			
BTC	Brain-Type Communications			
D2D	Device-to-Device			
eMBB	Enhanced Mobile Broadband			
HTC	Human-Type Communication			
IC	Integrated Circuit			
IoBNT	Internet of Bio-Nano Things			
IoT	Internet of Things			
IRS	Intelligent Reflecting Surface			
ISN	Intelligent Sensor Network			
LFP	Local Field Potentials			
MIMO	Multiple Input Multiple Output			
mMTC	Massive Machine-Type Communication			
MTC	Machine-Type Communication			
QoE	Quality-of-Experience			

This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

QoPE	Quality-of-Physical-Experience
QoS	Quality-of-Service
SNR	Signal-to-Noise Ratio
URLLC	Ultra Reliable Low Latency Communication
VoI	Value of Information
XR	Extended Reality.

I. INTRODUCTION

HE LAST two decades have witnessed tremendous new developments in information and communication technologies, most remarkably in wireless communications and Artificial Intelligence (AI). At the same time, the scientific understanding of the nervous system and the brain has also grown substantially. In fact, brain research is seen as arguably the most anticipated field of research for the coming decade. This is not a historical coincidence: the evolution of both domains is strongly interlinked. For example, on the one hand, the steep growth rates of technological advances in sensors, digital processing, and computational models have always supported the research in neurosciences while, on the other hand, the knowledge of how neurons and the neurological system work have supported the development of computational methods based on the Artificial Neural Network (ANN) [1]. A comprehensive review of the topic can be found in [2].

Neurosciences and wireless communications are converging in the context of several recent wireless and AI developments, where both are going to the edge: wireless communications is quickly heading toward nano communication while AI is moving toward edge intelligence at the sensor itself based on neuromorphic computing and various edge AI techniques, such as federated learning [3]-[7]. Futuristic technological solutions like Neuralink's novel brain implant [8] or the Internet of Brains [9] are perfect illustrations of potential opportunities ahead. While the former focuses on developing high-density, invasive wireless brain implants for humans, including a neurosurgical robotic system to insert the device, the latter provides the first experimental demonstration of a network of interconnected rat brains, configured as an organic computer, that outperforms single brain in behavioral tasks. Both approaches point at a future where human brains are part of the communications grid, interacting directly and seamlessly with other man-made devices but also with other brains. In fact, the ideas behind these technologies are strongly aligned with the vision of 6G [10], which is expected within ten years from now. We are aware that 6G is far from being standardized, and thus, current works may be highly speculative; in any case, the key cases for 6G are actually being defined now as illustrated by the eleven white papers recently published by the 6Genesis Flagship [11].

In this sense, this paper argues that one of the key drivers of future wireless technologies, such as 6G, would probably be wireless brain-machine interactions based on Brain-Machine Interface (BMI), enabled by a mobile network designed to support a new type of service that we call Brain-Type Communications (BTC), which can have many contrasts and synergies with the human- and machine-type communications of previous and current network technology generations (4G and 5G, respectively).

BTC would allow for more direct interactions between users and networks as compared with current systems, which are dominantly mediated by smartphones. Furthermore, BMIs are severely limited by wired communications, given that sophisticated applications should consider groups of individuals, each with implants made of thousands of recording channels, all part of a naturalistic scenario. New services supported by wireless BMI, such as interacting with the environment with gestures, motor intentions, or emotion-driven devices, impose remarkably different performance requirements from the current 5G in terms of Quality-of-Physical-Experience (QoPE). The list of applications is extensive, to cite but a couple of examples: wireless-BMI-connected intelligent vehicles, neural-based wireless networks with sensors and actuators working as an "artificial brain", as well as the future evolution of virtual reality services [7], [12], [13].

The main contribution of this paper is a novel, holistic tutorial that focuses on this new, promising research field located at the frontier of the two established domains: Neurosciences and Wireless Communications. Our goal here is to provide a tutorial of the state-of-the-art of those fields, mapping the most relevant activities and how they have a great potential to converge with the definition of the homocentric 6G [14] through the development of BTC. To this end, we expect to contribute to the ongoing discussion about the key wireless communications applications, in particular 6G, which will then affect the standardization process. In particular, we delineate the foreseen future applications and their challenges in two threads: **Neurosciences for Wireless Networks** and **Wireless Networks for Neurosciences**.

The first one refers to how current and new scientific/technological developments arisen from neurosciences can be employed as part of wireless systems as, for instance, direct wireless brain implants. The second topic refers to how wireless communications theory and technologies (mainly 6G) can support research and technological development in neurosciences. Topics in this thread include considerations of how communications/information theory can provide the fundamental limits of neuronal communications, which have a chaotic nature. We also present a case study—Brain-Controlled Vehicles (BCV)—that we have identified as an illustrative application that would benefit from the proposed merger between wireless communications and neurosciences. Finally, we discuss the security, privacy, and ethical issues underlying both topics.

Fig. 1 presents the key ideas and topics covered by this paper, mapping the future relations between wireless communications and neurosciences. We envision an interplay between the two topics supporting the development of BTC and widespread BMIs integrated with nanotechnology, among others. Based on the current trends and the scientific development to be presented in this tutorial, we argue that the encounter between these two fields has a great potential to already take place in 6G.



Fig. 1. Illustration of the proposed contribution along two threads: Neurosciences for Wireless Networks and Wireless Networks for Neurosciences around the concept of BTC.

In summary, we expect that this contribution can pave the way to a fruitful collaboration between researchers active in brain research, complexity sciences, and wireless communications to support in a timely manner the necessary activities to include applications related to neurosciences in the 6G standardization discussions that are expected to take place during the upcoming years. In addition, the paper will provide a single reference that symbiotically integrates the rather disparate state-of-the-art contributions in these two fields.

The rest of this paper is organized as follows. Section II provides the required background of neurosciences and brain research, specially discussing how brain signals are expected to be part of future wireless systems. Section III describes how neurosciences are contributing to the development of the next generation of communication systems through BTC, also providing details and challenges of wireless brain implants. Section IV presents the potential advantages that wireless communications may provide to neurosciences, considering the potential new generation of BMIs based on wireless connectivity for BTC and even the Internet of Bio-Nano Things (IoBNT), as well as theoretical and practical approaches related to the chaotic nature of neuronal communications. Section V introduces BCV as an existing application that would greatly benefit from synergistic research of 6G (or other future wireless communication technology) and neurosciences as proposed here. Section VI discusses the security, privacy, and ethical issues that are fundamental to guide both wireless communications and neurosciences in the near future. Section VII summarizes the paper pointing out our perspective for future research and technological development.

II. BACKGROUND

Evolution has shaped the animal brain to provide individuals with rapid, robust responses to multisensory, possibly conflicting stimuli, thus ensuring survival. We begin this section by highlighting brain design principles, with the focus on properties with direct relevance to wireless systems. Then, we describe three types of neural signals found in most of electrophysiological works, with an emphasis on spikes. We proceed by describing current implant technology for interfacing with the brain and then conclude with a review of the key concepts and current stage of BMIs.

A. Brain Design Principles

The brain is a complex organ, notably composed of nerve cells (neurons) but also of supportive cells, such as glia. Ultimately, one may attribute the diverse components, structures, and dynamics found in brains from different species [15], [16] to singular evolutionary pressures.

Brain regions that are mainly made up of electrically insulated neuron axons (myelinated) are referred to as white matter, whereas neuronal cell bodies are found in the gray matter. From a communication systems perspective, white matter may be seen as insulated wires connecting widespread neural populations in the gray matter. The probability of two cortical neurons being connected is 1 in 100 within a vertical column of 1 mm in diameter, and 1 in 1000000 for distant neurons; further, forty to sixty percent of the brain mass volume is due to wiring (for comparison, the volume fraction of wiring in a computer microchip may reach up to 90%), and only one quarter of all energy is spent by white matter [17]. The brain presents local, densely connected neural populations that are sparsely connected often with small-world properties. A direct consequence of such a connectivity pattern is a disproportionate increase in the white matter (wiring) volume as cortical gray matter increases. This poses great challenges for wireless neural recording technologies.

Evolutionary optimization of neural connectivity is certainly constrained by energy consumption. Nearly half of the brain energy consumption is due to spiking activity, arguably the essential method by which neural populations communicate [17]. Single neurons have a physiological upper limit on firing rate in the range of hundreds of Hz [18], leading to a potential bandwidth of a few Terabits/s for the whole brain. This limit, however, is never reached because of the energy limitation. Considering the human brain metabolism, the average spike rate can be no higher than 1 spike per second per neuron [19]. Thus, communication systems operating with BTC protocols should be aware of these natural brain bandwidth limits.

The locally dense, globally sparse connectivity scheme constrained by the brain energy budget may reduce the signalto-noise ratio [20]. Considering that more reliable neurons would require a superlinear increase in energy cost (caused by neuron physiology), one alternative is to average out large numbers of (noisy) neurons. But that, in turn, would possibly lead to redundant neuronal activity, which is not energyefficient, unless the network is able to reconfigure on the fly, suppressing connections that contribute little to good choices and reinforcing (making more efficient) those that do not. This overly simplified description is known as neural plasticity, the capacity of neural networks to modify their connectivity patterns based on correlated neural activity and behavioral feedback [21]. In summary, learning from experiencing the world to optimize behavior is a central mechanism that supports brain design principles under a limited energy budget. The immediate impact on the development of new wireless networks is that technologies underlying BTC must be adaptive to accompany brain plasticity and preserve brain learning mechanisms.

B. Neural Signals

In this section, we describe three types of neural signals that compose most of electrophysiological works [22] and that are central to BTC systems. Considering invasive recording methods, spikes relate to the rise and fall of the membrane potential of a single neuron over time [23]. Neurons are essentially formed by the soma (the cell body), dendrites, and an axon. Typically, electrical and chemical signals are received in the dendrites and soma of the neuron, whereas the neuron axon transmits an electrical signal to other neurons. Neuronal communication is mediated by synapses, in which we observe the propagation of neurotransmitters through the space between neurons (synaptic cleft). Along the neuronal axon there are voltage-gated ion channels, which regulate the ionic current flow as a response to input signals to the neuron. The rapid difference in ionic concentration inside and outside the cell originate the action potential (spike), which flows over the axon and targets other neurons. This signal has a strong nonlinear dynamics (Fig. 2) owing to neuron physiology.

Spikes are normally sampled at 40 kHz by multielectrode arrays, each electrode capturing the resultant membrane potential of the surrounding neurons. This multidimensional signal



Fig. 2. Representation of the most common brain signals used in BMI: noninvasive EEG (top right panel), invasive LFP, and spike (middle and bottom panels). Electrocorticography (ECoG) signal properties are similar to those of EEG; however, because it is an invasive method, it is far less used in human studies. Figure adapted from [49].

is then fed into a spike-sorting algorithm, responsible for identifying the membrane potential time series of each individual neuron [24]. Next, spike times are identified and saved either as a time stamp vector (millisecond resolution) or as a binary vector (1 if a spike has occurred, 0 otherwise). The sequence of spikes over time from a single neuron is known as a spike train, which is the data structure used as the input to the spike-based BMIs [25].

The same time series used to construct spike trains can be used to extract another signal, the Local Field Potentials (LFP). For that, a low-pass filter (<300 Hz) is applied to the raw electrode signal and then downsampled, usually to 1 kHz. LFP relate to the superimposing electrical potential of thousands of neurons surrounding the recording electrode [22]. The spectral power density of this field is inversely proportional to frequency and is transmitted through brain tissue, a phenomenon known as volume conduction. The most common input in LFP-based BMIs is features extracted from the LFP frequency power spectrum [26].

The typical signal used in noninvasive approaches is the electroencephalography (EEG). EEG and LFP oscillations share similarities [22], [27], but, because the recording electrodes are further away from neuronal sources, noise, muscle contraction artifacts, and other tissue-related interference make EEG a less effective signal than invasive recordings. EEG is commonly recorded from 16 to 128 channels, studied at frequencies up to 100 Hz, and thus, the sampling rates rarely exceed 1 kHz.

C. Neural Interface Technology

For the purposes of wireless communications and BTC systems, we will focus on invasive signals. The main advantage of being invasive is the closer interface with brain cells, which leads to less noisy, more reliable readings with a more granular information content, a fundamental feature for complex applications [28], [29]. Novel signal processing algorithms have a substantially increased noninvasive (EEG) BMI performance [30], whereas standard brain implants present physical and longevity issues [31]. Nonetheless, there is a general agreement that, once technological and ethical issues have been addressed, invasive BMIs should prevail in the vast majority of applications [32], [33]. Brain implants and recording technologies have been developed for decades now. They are generally composed of six different parts: a probe, epoxy fill, an acquisition Integrated Circuit (IC), a circuit board, connectors, and an external cable [34]. The interface with the brain tissue is actually solely the probe, from which the signal travels from or to the acquisition IC that sits on the circuit board through the epoxy fill and the connector. The cable connects the whole structure, fixed on the skull, to the external supporting hardware.

One of the state-of-the-art devices regarding invasive implants are the neuropixel probes [35]. For instance, the advanced CMOS microelectronics and small, lightweight shanks allow for eight probes simultaneously implanted in a mouse brain, and more than 3000 recording sites. For each probe (384 recording channels), data acquisition is estimated at 1 GB/min at a sampling rate of 30 kHz. Considering that the mouse brain is orders of magnitude smaller than the human brain, these recording facts clearly highlight the substantial requirements in data processing and recording in the context of BMI and modern wireless communications.

The main challenges regarding invasive devices are related to their implantation procedure (sometimes through open skull surgery), biocompatibility, and longevity [36]. Implant functionality is impaired by tissue scarring as well as foreign body reactions that promote the degradation or breakage of the probes [37]. Furthermore, fully immersed implantables that rely on tethers to connect the device to an external interrogator inhibit long-term usage and reliability as this connection can be broken easily through movement or patient activity [38]. Tethers have additional challenges including the lack of scalability and a greater body reaction. The number of neuron interface channels is also limited to the number of tethers, and even though the relationship is not direct as one tether can have many probes, they are not a good choice when multiple areas of the brain are planned to be interfaced with a single system. On top of that, as the targeted area of study is deep in the brain, this will result in larger tethers that are harder to manage. Thus, the option of a wireless-based system has raised the interests of many researchers in the area (see [39] for a review).

Many technological breakthroughs will be required before functioning wireless BMIs will be widely available [34]. To begin with, wireless-based implants have to account for the many barriers imposed by the brain in order to be functional [28] (these barriers will be explained in more detail later in Section III-B). Wireless devices also need to interface with neurons whilst having the capability of converting wireless energy into circuit current (batteryless). This added complexity is nowadays feasible with energy-converting mechanisms based on microelectronics and nanotechnology [39]. Additionally, wireless implants require consideration of the human body as a communication channel. Because the brain is comprised of multiple different tissue types, and each type poses different interactions with the propagated waves, the wireless communication system between implantable and external devices, or derivations thereof, must precisely choose the frequency range and operating mode [40]. For example, while brain-stimulating devices for epilepsy require stimulation in random short-term periods, for Parkinson's disease the stimulation is constant at a particular rate at different times. The same goes for sensing applications.

Despite the challenges, the development of wirelessbased neural interface technology will support the study of freely moving animals away from highly controlled laboratories [41]–[44], which will transform research in neurosciences considering that cognitive processes emerge from brain–body– environment interactions [45]. In parallel, BTC systems may underlie the next wireless communications revolution. As we argue throughout the paper, there is an intricate but feasible technological, medical, and ethical path ahead.

D. Brain-Machine Interfaces

The rapid progress in neural recording technology has paved the way for the development of BMIs [25], [46]. A BMI is a closed-loop framework, in which neural signals are sampled, preprocessed, and fed into a decoding algorithm (regression or classification) that can map behavioral intents from the brain to artificial devices, whose action outcomes are perceived by the subject sensory systems, thereby closing the loop. Applications are diverse, from shedding light into basic neurosciences research [47] to contributing to motor rehabilitation in spinal-cord-injured patients [48].

From a technological perspective, BMIs rely on the continuous progress in electrode design [50], data recording [51], and signal processing [52]. However, there is a central gap in the BMI research that is shared by other fields of neuroscience: what is the essence of the neural code? In other words, what are the features of neural activity that carry information about sensory stimuli and cognitive behavior? For instance, there is solid evidence favoring rate and temporal codes [53]–[55], but it remains unclear what the anatomo-neurophysiological patterns and information processing mechanisms of such codes exactly are.

For spike-based BMIs, most decoding algorithms map the spike rate or inter-spike time interval changes into behavioral choices. The rationale is that motor and cognitive acts (or intentions) modulate single neuron responses in diverse brain regions, and thus, spikes carry sensory and task-related information. As single neuron responses vary considerably within and between task trials [56], the use of recordings from populations of neurons results in more robust interfaces [57]. The common target for brain implants is the cortex region, the outer layer of the brain, from which sensory and motor information have been successfully extracted [25]; nevertheless, deeper brain regions, such as the basal ganglia and the cerebellum, have a fundamental role in action selection and spatial localization, among other important aspects of animal behavior [23], but are harder to be reached safely.

If, instead, LFP signals are to be used, the common approach is to extract frequency power spectrum features from data blocks over time as the behavioral task unfolds [26], given that specific frequency bands have been shown to correlate with behavior [58], a fact that also holds for EEG studies. Finally, neural plasticity is fundamental for BMIs to operate properly [47], [59], [60]. BMI design has to carefully consider processing time and sensory feedback delays, and thus, modern communication systems have plenty to contribute.

E. Wireless Communications and the Brain

As discussed above, neural signals and their application in BMI pose new challenges for wireless networks, mainly for the established communication systems that are designed to support transmissions related to humans and/or machines, not brains.

From 2G and 3G to the iPhone generation, applications have focused on *connecting people* in a variety of ways; hence Human-Type Communication (HTC). However, the past decade ushered in a whole new type of wireless communication dedicated to connecting machines within the Internet of Things (IoT) system. Indeed, the emergence of Machine-Type Communication (MTC) links has revolutionized the wireless industry and is the driving force behind the ongoing deployment of 5G wireless systems. At this juncture, it is natural to pose the following question: What types of mobile devices will disrupt the wireless industry and drive beyond 5G wireless systems in the same manner that the iPhone and the IoT have done?

Although a conclusive answer to this question is not possible at this time, it is very natural to posit that next-generation wireless devices will no longer be handheld smartphones or IoT sensors in the field, but they will rather be wearable devices along a human body, including human brain implants. This observation is not a mere speculation, but it is instead motivated by the tremendous advances that we are witnessing in the area of wearable and human-embedded devices. Neuralink's recent achievements are a prime example: a public demonstration in late August 2020 showed a successful wireless BMI in pigs using a miniature, bluetooth-based device with 1024 channels that serve both for recording and stimulating the brain. In addition, the shift toward implants is further motivated by several emerging wireless services, such as immersive Extended Reality (XR) and BMI, in which the human body and brain become an integral part of the wireless service [10]. In these services, it will soon become necessary to provide communication links among not only machines (MTC) and human users (HTC) but also among the brains of different users. Hence, we foresee that BTC will be the next frontier in wireless connectivity, as indicated in Fig. 3.

The main lesson learned in this section is related to the very specific type of signaling that constitutes braincommunication-based neuronal activity, which poses enormous challenges for communication engineers. In specific terms, BTC links must be designed in a way to seamlessly connect a human brain to a wireless network and potentially provide two-way communication among the users' brain implants and the various networks and IoT devices. A unique feature of BTC links is that they will require the network to match the operating functional complexity of the human brain with a given application of choice. In the next section, we highlight how the brain's inherent features can soon



Fig. 3. Illustration of different types of BTC services in wireless networks. The figure highlights the communication between brain implants via Uplink/Downlink BTC services and a base station, or directly, via brain-to-brain BTC links between wireless implants.

become an integral component of wireless networks that cannot be ignored when modeling, analyzing, and optimizing the wireless networks of the future.

III. NEUROSCIENCES FOR WIRELESS NETWORKS

In this section, we will discuss how the technological developments based on the state-of-the-art in neurosciences will open many new opportunities with related challenges in wireless communications beyond 5G systems. This will include the support of BTC and intelligent (neuromorphic) sensor networks based on spikes.

A. Direct Brain Implants That Communicate Wirelessly

Communications with brain implants will be a hallmark of next-generation wireless networks, and hence, we must have a deeper understanding of how to deliver wireless services to networks with *brain-in-the-loop*. To do so, we will first discuss some use cases that highlight different ways in which BTC will be integrated in wireless networks. Then, we delve into the various challenges associated with the identified use cases and conclude with discussions of open research problems and some preliminary results.

1) Use Cases: The first step toward understanding the unique wireless challenges of BTC links consists of delineating possible BTC use cases in an actual network. In this context, we envision three key use cases (as illustrated in Fig. 3).

- *Downlink BTC:* BTC links can be used in the downlink of a wireless network. Here, the downlink transmission links are used to transmit data from the network toward brain implants. A chief use case in this context is XR services. Indeed, next-generation XR services may tap directly into the human cognition in order to provide a truly immersive virtual world where a wireless user can navigate using brain-based signals along with various body-implanted sensors. In such use cases, the brain is the receiver of the wireless data, and thus, the downlink BTC traffic will require high data rates.
- Uplink BTC: BTC links can be used for uplink communications in order to transmit information extracted from

the human brain through its implants to other network devices and servers. Two key BMI examples that require uplink BTC are multi-brain-controlled cinema [61] and wireless cognition [62]. The first relates to hundreds of spectators interacting, through brain input, with an audiovisual performance that unfolds based on their emotions, reactions, and cognitive engagement. In this scenario, participants may interact among themselves and with the performance, creating a unique experience and possibly revolutionizing the entertainment industry. The latter depicts a scenario in which a drone or an autonomous vehicle is controlled by one or multiple brains (see Section V for details). The two examples highlight critical applications that require multiple, reliable recordings with several brain regions, including deep areas, such as amygdala (emotions) and cerebellum (motor control), which have to be processed to deliver real-time responses to the target application. Hence, uplink BTC is fundamental.

Brain-to-Brain Communications: BTC links can be used to establish direct communications among the brain implants of different users within the same or different environments [9]. Brain-to-brain communications can be seen as the next step in Device-to-Device (D2D) communication, in which the devices are now direct brain implants. Brain-to-brain BTC links can be useful in many scenarios, such as immersive gaming, creating unique user interaction scenarios, and education, by enhancing the possibilities with which cooperative problem-solving activities can be developed with groups of students (social brain networks). In this latter scenario, there is also substantial evidence that the synchronization of brain rhythms, both individually and in a group, is related to learning. Thus, brain-to-brain links could be used to bolster student engagement as part of effective teaching methods.

2) Challenges: Having laid out the key use cases for BTC, our next step is to identify the unique challenges of these use cases, compared with traditional HTC and MTC services. First, it is well known that the bottleneck of HTC services is downlink communication, whereas the bottleneck of MTC services is uplink communication. In contrast, in BTC, we can easily see that both uplink and downlink may constitute a bottleneck for data rates. On the one hand, to provide immersive experiences, significant data must be downloaded in the downlink toward the brain implants. On the other hand, in order to provide sensory and control inputs from the human brain to the network and its services, brain data must be transmitted from the implant to the network. At first glance, one would think that the uplink input will still be short packet, small data, as is the case for MTC. However, results in [62] show that the amount of data generated by a brain for wireless cognition services can be in the order of terabytes. Hence, uplink BTC will also require ultrahigh speeds from the wireless links. which is in sharp contrast with MTC services.

Second, despite its immense computational abilities, the human brain has its own perceptual and cognitive limitations. These cognitive limitations can be affected by multiple human brain sources such as context, attention, fatigue, or limited cognitive abilities. From a wireless perspective, these cognitive limitations can be translated into limitations on the way in which a human brain perceives network Quality-of-Service (QoS) metrics, such as rate or delay. For example, as shown in [63], because of its architecture and neural network dynamics, the brain may exhibit intrinsic time delays that affect the way in which it perceives the world around it. Therefore, a key challenge here is to develop new techniques from neurosciences in order to provide new models for the brain that can quantify these limitations and potentially be used in a wireless network framework to map those limitations into QoS or Quality-of-Experience (QoE) metrics. Note that this challenge differs here significantly from traditional QoE metrics, such as the mean opinion, in which one can simply use interviews or basic experiments to quantify QoE. Instead, here we need to quantify the QoPE introduced in [10], in which the specifics of a human's physiological characteristics, particularly the brain, must be captured and mapped into conventional wireless QoS metrics. Note, however, that the actual requirements for BTC will be defined by the specific applications, and thus, it is unfeasible at this point to quantitatively define their minimum quality levels.

Third, 5G systems are expected to deliver three broad types of services: Enhanced Mobile Broadband (eMBB) services, in which high data rates are expected, Ultra Reliable Low Latency Communication (URLLC) services, in which reliable low latency transmissions are required for services such as IoT sensing that do not require high rates, and Massive Machine-Type Communication (mMTC) that deals with the connectivity of a massive number of IoT devices. Traditionally, these service classes are expected to be distinct from one another. For example, URLLC services are assumed to not require any data rate guarantees because they deal with short-packet transmissions of IoT sensor data. Meanwhile, eMBB services simply require a high rate and do not need much reliability or low latency guarantees. In contrast to these traditional service classes, BTC services may require, simultaneously, high reliability, low latency, and high (eMBB-level) rates. For example, wireless cognition and remotely controlling an autonomous vehicle by the brain may call for very high reliability and very low latency because of the criticality of the circulating data. Meanwhile, this remote control may also require very high rates as discussed in [62]. Hence, when dealing with some BTC services, it is potentially necessary to provide both eMBB-level rates and URLLC reliability and latency, which is yet another key challenge. Moreover, as the technology becomes more mainstream, we can anticipate massive numbers of BTC links active at a given time, and hence, in this case, mMTC features will also appear, particularly for brain-to-brain links. Clearly, the evolution toward BTC may require us to revisit the existing 5G distinction among different services. Nevertheless, a detailed comparison between BTC and the 5G-defined regimes URLLC, mMTC, and eMBB is not vet possible because it will depend on the specific definition of the BTC applications.

Fourth, although brain-to-brain BTC links share many of the aforementioned challenges, they also bring a new dimension that has to do with the interactions among human brains, which have different physiologies and cognitive capabilities. Addressing this challenge requires a better understanding of networks of brains and how they may interact with one another. Naturally, brain-to-brain communications brings in a suite of interdisciplinary challenges that require a better understanding of not only the communication features of brain-to-brain, but also the potential interactions among the brains of different users whose context, demographics, and characteristics are disparate.

A qualitative comparison between HTC, MTC, and BTC is presented in Table I. The table provides a high-level comparison between these three communication paradigms. Although the numbers for HTC and MTC have been defined by the standardization bodies, it is unfeasible for us now to numerically determine the different applications of BTC together with their particular quality requirements. Our objective is rather to indicate that BTC will open a new paradigm for diverse applications that may potentially include services that are more demanding than those defined for 5G and being under discussion for 6G.

3) Research Problems: Clearly, the aforementioned challenges bring forward interesting research problems at the intersection of neurosciences and wireless networks. In general, providing wireless networking with "brain-in-the-loop" is a rich research area with many open problems that follow directly from the identified challenges.

One of the first open problems in this area pertains to the need for new techniques that combine neurosciences with wireless network modeling in order to precisely quantify QoPE measures. On the one hand, one can take a data-driven approach to this problem and look for new machine learning techniques that can dynamically build QoPE metrics by learning from the network users and their brain behavior. Naturally, the primary limitation of this approach is that it will require significant datasets and long-term observation. However, as datasets in both the neurosciences and wireless communities are becoming more accessible, we anticipate new opportunities for designing QoPEs. On the other hand, one can forego the data-driven approach or complement it with an analytically rigorous approach to model the brain's features. In particular, one can leverage existing tools from control theory and neurosciences to view the brain as a control system with a feedback loop and, then, use this observation to quantify how different inputs (from the wireless network) are translated into meaningful information for the brain. We can potentially study the transfer function of this brain control system and understand its behavior with respect to different input excitations coming from a wireless network. By using this approach, we can potentially investigate how QoS metrics are translated into QoPE. This can benefit from some of the existing studies on how to look at the brain's control signals (e.g., see review in [65]). Last, but not least, real-world experiments with actual participants can be organized to better understand how the brain perceives QoS. These behavioral experiments can be combined with behavioral frameworks, such as prospect theory and cognitive hierarchy theory [66]–[69], that explain how humans make decisions to yield new insights into how

to model the response of a brain to wireless signal inputs for different services.

Moreover, as discussed above, there is a need to calculate the processing power of the brain, using techniques of neuroscience, so as to truly quantify the amount of data needed. Here, instead of looking at brain limitations, we are more interested in the brain *capabilities* and how these can impact wireless communication. While the calculation of [62] provides a first step in this direction, there is a need for more rigorous modeling that takes into account realistic brain models or real-world brain data.

Once QoPE metrics are developed and brain capabilities are quantified, a very natural next step is to investigate how network management, multiple access, and network optimization techniques will change when dealing with BTC links and QoPE. In particular, one can design new brain-aware resource management techniques that can tailor the network resources and operation to match the performance required of the brain while also being cognizant of the brain capabilities as well as its inherent limitations in processing information, in general, and processing wireless QoS metrics, in particular. One fundamental question that we can pose in this area pertains to whether or not brain constraints lead to a "waste" of wireless resources because of the delivery of a QoS metric that cannot be perceived by the brain. For example, it is natural to ask whether a human brain can see a difference between two different delay values, i.e., will 10 ms be perceived as a better QoS than 20 ms?

Moreover, the coexistence of BTC, HTC, and MTC links, which is expected in early-on deployments of beyond 5G cellular systems, will bring forth a rich set of resource management questions pertaining to how one can enable a seamless coexistence of these fundamentally different service classes, as presented in Table I. Here, beyond investigating radio resource management problems, we can also investigate new ways to incorporate brain features into network slicing problems. Indeed, network slicing must now handle a new type of service, and hence, a rich set of new open problems can be observed. Furthermore, because BTC links carry characteristics of all three traditional 5G services, i.e., eMBB, URLLC, and mMTC, it is necessary to investigate how one can guarantee a high rate, low latency, and high reliability simultaneously, in the presence of a potentially large number of BTC links. Here, one can start by first identifying the achievable performance of BTC links over 5G and beyond systems (e.g., over terahertz or millimeter wave systems). In particular, there is a need to analyze the rate-reliability-latency operation regime that can come out of the deployment of BTC links over a cellular system and then translate this analysis into a feasible QoPE regime of operation that maps the ratereliability-latency requirements into QoPE measures. Once this feasible QoPE regime is identified, one can revisit traditional problems of multiple access in order to see how all three factors-reliability, latency, and rate-can be matched to the requirement of both the user's brain and the network service that is being adopted. Indeed, here one important direction is to study how different types of services (e.g., XR, BMI) will have different brain and QoPE requirements.

Requirements/features	HTC over cellular	MTC over cellular	BTC over cellular
Uplink	High signaling overhead to overcome features such as mobility (handover). Relaxed/mild throughput requirements.	Reduce signaling over- head (e.g., via fast uplink grant [64]) and specialized random access procedures to support high user den- sity. Characterized to con- sider short package–small data.	Brain control BTC may require reliable and low-latency transmission with varying throughput (low to high). Some special cases have high throughput requirements with high signaling overhead to support mobility.
Downlink	High throughput require- ment for individual users.	Most typical applications demand low throughput for individual users, which is mostly used for soft- ware update/upgrade of the embedded systems.	High throughput require- ments with high signaling overhead to support mo- bility.
Subscriber load	Few users that are mostly supported by small cells.	High density of devices	High user density is ex- pected with URLLC and high data rates simultane- ously.
Device Types	Broadband devices, such as smartphones, tablets, personal computers.	Devices are mostly small (IoT) devices carrying sensors with specific power constraints.	Brain implants (invasive and noninvasive).
Delay requirements	Vary between best-effort services as e-mails, and high-definition games to support human sensitivity in terms of latency.	Many delay-tolerant applications and also low-latency ones, the last mostly employed in closed-loop control systems, protection systems, and other critical applications.	Strict delay requirements for both downlink and up- link.
Energy requirements	Relatively high energy consumption because it supports many applications that run on a smart phone. In most of the cases the battery is charged once per day on average.	MTC devices usually perform specific tasks that provide flexibility to set up the data transmission duty cycle. In many cases, the battery should be recharged/replaced in long periods of time, e.g., 5 years.	High energy efficiency is needed to support a high duty cycle demand, such as the case of brain-to- brain communication.
Signaling requirements	Accurate signaling proto- cols are required because of uncertainties related to mobility and data usabil- ity.	Reduced requirements in UL in order to support massive numbers of users.	High level of signaling for both uplink and down- link is required to support URLLC, high data rates, and massive numbers of users simultaneously.

TABLE I

MTC VERSUS HTC VERSUS BTC FEATURES AND REQUIREMENTS IN THE CONTEXT OF MODERN WIRELESS NETWORKS

In this context, 5G and modern wireless technologies are essential to support mobility requirements, not only in providing high rates of data traffic but also ensuring security and privacy. It is often accepted that 5G can reliably operate up to 500 km/h, and 6G twice that velocity [70], which would facilitate coverage in high-speed trains and airplanes. Further, it is anticipated that modern wireless technologies, such as 6G, will accommodate the current conflict between high date rates and high mobility with advanced handover policies and integration of heterogeneous networks [14].

Another important open problem is to quantify and measure information from brain implants. Here, information is no longer standard digital information, but instead a byproduct of a user's brain. Hence, using tools from fields such as information theory, we must study how "information" can be modeled when it is the output of a brain (e.g., using information-theoretic perspectives). Then, we can revisit the recently introduced concepts of Age of Information (AoI) [71]–[73] and Value of Information (VoI) and see how these metrics change when dealing with a brain network. For example, we can observe that the way in which information "ages" when it is transmitted among brains may no longer be linear, as is the case for traditional wireless information transmission. In this respect, aging of brain information transmitted over BTC or brain-to-brain links may require new approaches that depart from the classical linear aging process that is used in most of the AoI literature. Here, it is necessary to investigate how information propagates in a brain (e.g., using models such as those in [63]) to see how timing delays and the neural composition of the brain capture and process information. Similarly, new ways to quantify VoI when it is the output of a brain are needed. Once information is quantified and its different metrics are revisited, we can leverage this analysis for both physical layer designs as well as for routing and information flow problems, as discussed next.

At the physical layer, the deployment of BTC will require new designs. Here, we anticipate a need for merging tools from neurosciences, information theory, and communication theory. To this end, we must investigate how the brain information that is quantified using information theory techniques can be translated into digital waveforms. For example, here, by taking the control-theoretic approach for modeling the brain, we must investigate how the output of the control system model of the brain can be translated into a digital communication signal that can be transmitted over BTC links.

Finally, in terms of routing and information flow, it is necessary to develop new techniques to manage brain-to-brain and BTC links in a way to optimize AoI and VoI metrics when those metrics pertain to a brain. As already discussed, the models for these two metrics will be significantly different when dealing with human brains. As such, existing latency-optimal or rate-optimal routing and information flow algorithms will not necessarily be AoI-optimal or VoI-optimal. Hence, we envision many fundamental routing problems that can now see a wireless network as an overlay of two interrelated systems: a) a human-to-human network that receives and translates information through a brain and b) a D2D network that carries this information. Modeling the relationship between these two systems and integrating it into network routing and information flow optimization problems is clearly an important and meaningful open problem that brings together neurosciences, communication theory, and network science.

4) Sample Results: The area of wireless network design for incorporating BTCs is still in its infancy, and hence, not many works have looked at related problems. However, in [74], we have made the first step in this direction by analyzing how to use a data-driven approach that uses user brain information to create QoPE measures that map wireless delays into perceptions of the brain and, then, those perceptions are integrated into a resource allocation problem. In this early work [74], to model QoPE, we explored the observation in [63] that the brain can have multiple "modes" depending on the age, sex, demographic, time of day, and other social features, to learn how the brain perceives delay in a wireless network. The QoPE therein pertains to how one can translate a brain mode (extracted from the data) into a perception of QoS metrics, such as delay. Our work in [74] showed that, for a wireless network, the aforementioned brain mode limitations make the wireless user unable to distinguish the QoS differences between different wireless delays. In other words, the QoPE of a user maps each delay value to a different brain perception. Building on this observation, our results in [74] show that because of the cognitive limitations of the brain, delivering ultralow latency for services such as XR may not improve the user experience, because at a very low latency regime, the user's brain can no longer distinguish the difference between different delays. For instance, our results show that it is less probable that a user distinguishes between a 20 ms and a 10 ms delay compared with 30 milliseconds and 20 ms. As such, when designing BTC links, one key challenge is to properly model and capture the



Fig. 4. Early result from [74] showing how a brain-aware resource management approach can save significant resources (in terms of power), particularly in the low latency regime, by being aware of the cognitive limitations of a brain that limit its perception of delay. Human cognition is physiologically limited in terms of sensory perception and motor control, and thus, there is a delay threshold in the sensorimotor loop below which the human brain cannot perceive any improvement in QoS. The x-axis represents here the "raw" maximum tolerable delay threshold of each user.

limitations of the brain and factor in those limitations into the wireless network design.

In addition, in [74], we then incorporated the learned brain limitations into a downlink power control problem with brain perception constraints. We did so in order to test the hypothesis that the brain-aware resource allocation approach can significantly save network resources. Here, we have particularly shown that, by explicitly accounting for the cognitive limitations of a human user's brain, i.e., the minimum delay below which users may not be able to perceive because of sensorimotor physiological limitations, the network can better distribute resources to BTC users that need it when they can actually use it. To illustrate this, let us consider an interactive and immersive gaming scenario. No human is able to perceive image flickering with frame rates above 48 Hz [75], whereas motor control is in the range of a few hundred milliseconds [76]. Thus, there are minimum visual frame rates and body tracking sampling rates above which the human brain cannot perceive any improvement in QoS. This is in stark contrast to conventional brain-agnostic network resource allocation techniques in which resources may be wasted, as they are allocated only based on application QoS without being aware of whether the human user's brain can realistically process the raw QoS target of the actual service. For instance, in Fig. 4, extracted from our work in [74], we compare the performance, in terms of power allocated to optimize the wireless system while meeting the delay threshold and reliability constraints, between a brain-aware resource allocation approach and a brain-unaware resource allocation approach. Strikingly, this figure shows that at very low latencies (below 40 ms), a brain-aware approach can save significant resources by being aware of the fact that the brain of a user (depending on the mode of the user) may not distinguish a QoS difference between different values of latency. Clearly, these promising results can be used as a building block for new research in this area that can potentially address the rich set of open problems previously identified.

B. Brain Barriers for Wireless Channels

The physical medium also poses challenges to any wireless system that would support BTC. Remarkably, the transcranial wireless channel presents many challenges to the many wireless technology options because of its structure and function. The brain is covered by the skull and surrounding head tissue that absorb or scatter high-frequency signals.

Lower-frequency signals are known to require the implantation of large devices and may increase head heat. Novel wireless solutions must cope smartly with those unwanted effects, but we must take into account that single neurons are known to have high data rate demands for sensing purposes. We now explore the listed brain barriers in [28] focusing our discussion on the wireless technologies and on the communication channel between implants and external devices.

1) Spatial-Temporal Resolution: The number of neurons and other brain cell types goes beyond the billion unit mark and is considered the major challenge in measuring the whole brain information with the existing technology and infrastructure. Naive estimates of the lower bound data rate of the whole brain recording is about 100 Gbits/s, which is already a challenge for today's wireless technologies, let alone for future BMIs. The forthcoming technologies must include compression techniques that minimize the transmission burden of single action potentials. The compression technique will have an interesting interplay with the sampling rate of signal recording and the wireless technologies and their equivalent data rates.

2) Energy Dissipation: The propagation of transcranial wireless signals that are transduced by implantable devices will result in energy dissipated through the tissue. This energy will be converted into heat, which is also dissipated. Owing to the tightly packed structure of the brain, damage can occur as a result of a minimum temperature increase of above two degrees Celsius. Wireless signals, however, can be easily modulated in order to operate below the 100% duty cycle of the system operation, which can help prevent damage caused by energy dissipation. The brain also has natural cooling mechanisms that can help restore normal brain temperature. However, the real challenge lies in the large-scale deployment of heavy and dense recording and stimulation techniques for high spatial resolution.

3) Volume Displacement: Insertion of devices into the brain can cause an increase in its volume, leading to damage to its functioning tissue. Wireless technologies can help keep these implantables small by using high-frequency transmission that enables the decrease of the dimensions of the antenna elements. If low-frequency transmission is required, the devices will become larger with larger antenna elements making them unfeasible to implant into the brain tissue. Therefore these larger devices will possibly reside above the cortex, i.e., in the sub-dural brain space. Furthermore, glial scarring (formation of glial tissue around the implant preventing its interface with neurons) may inhibit the implantable from functioning properly. Wireless interfaces can help long-term implants by packaging devices within biocompatible material that prevents foreign body reaction. Future techniques, such as Multiple Input Multiple Output (MIMO) wireless systems for implants, might help the use of low-frequency solutions for deep brain interfacing that is essential for integrating existing wireless system platforms into future wireless brain interfaces.

C. Brain Implants Assisted by Intelligent Reflecting Surfaces

Despite the increased risk of injuries and other related issues, invasive wireless brain implants exhibit numerous benefits in comparison with conventional over-the-scalp solutions. It has been shown that these prosthetic devices are capable of sensing brain activity more accurately, interacting directly with the brain and providing a higher Signal-to-Noise Ratio (SNR) [77]. These capabilities make them powerful tools for enabling BMI in future generations of wireless communications (e.g., 6G and beyond). However, before this technology becomes available to the global population, many limiting issues need first to be addressed. One important limitation of wireless brain implants is related to the strong signal attenuation caused by tissue blockage and absorption. The high quantity of water molecules in the human body can interact with the electromagnetic waves, absorb a significant part of transmitted power, and distort the radiation pattern [78]. Such a characteristic can deteriorate the communication link and impact reliability. One could, to some extent, alleviate this issue by allocating a higher transmit power; however, this parameter cannot be increased indiscriminately. Firstly, there are strong power restrictions due to human health, and because radio frequency energy heats up brain tissue, there is a safety-specific absorption rate limit of 0.4 W/kg averaged over the whole body and a maximum output power density of 10 mW/cm² for electromagnetic waves passing through tissue [79], [80]. For illustration purposes, the total transmission power of a wireless neural recording system reported by Schwerdt et al. [81] was 47 mW, whereas a device based on ultrasound communication achieved 0.12 mW [44]. Secondly, in general, the implanted devices have limited access to energy resources. Therefore, new energy-efficient strategies for improving wireless transmission performance in brain applications are required.

In particular, Intelligent Reflecting Surfaces (IRSs) have recently arisen as appealing devices for smart control of the electromagnetic propagation environment. An IRS consists of a two-dimensional structure that comprises a large number of nearly passive subwavelength metamaterial elements with tunable electromagnetic properties. These elements can be dynamically configured to collectively change the behavior of impinging wavefronts so that capabilities like steering, polarization, filtering, and collimation can be achieved [82]. Such features make the IRS technology attractive for improving the performance of wireless communication in brain implants. Conceptually, if the prosthetic implanted device is assisted by an IRS, it could send and receive information more reliably without increasing its power consumption. This improved brain communication system could be implemented, for example, by implanting IRSs between the skull bone and the scalp for assisting sensors and actuators implanted deeper in the brain. In this architecture, the deeper implants would exchange information directly with the brain, while the IRSs would assist the wireless communication established with external devices. An IRS-assisted BMI can become able to provide the following capabilities.

- *Improved Reliability in wireless data transfer:* by proper tuning of the IRS elements, the signal transmitted from the brain implants can be boosted so that a higher SNR can be achieved at an external receiver, thereby improving the communication reliability.
- *Reduced power consumption in the brain implants:* because the SNR can be improved with the help of IRSs, one can decrease the transmit power at the brain implant and still achieve a satisfactory communication performance. This would reduce energy supply requirements and prolong the battery life of the implanted devices.
- *Improved communication security:* because implants can sense and stimulate the brain, security issues become a critical concern in BMI. An IRS can also be beneficial in this context: it can null out information leakage at a potential eavesdropper, or it can operate in the shield mode to avoid brain hacking; that is, by properly optimizing the IRS elements so that transmissions to the brain implants from a hacker can be completely absorbed/blocked.

All in all, the development of IRS technologies in the years to come combined with brain implants would support the development of BTC applications in the next generations of wireless systems.

D. Lessons Learned

Throughout this section, we have provided what we foresee as a new paradigm for a future wireless communications system: the brain-type communication (or simply BTC). The main differences from existing human- and machine-type communications are schematically presented in Table I. We have also put forth an argument that BTC has the potential to emerge as one of the main application domains of the forthcoming 6G. We have also described the challenges that the "brain environment" poses to the deployment of wireless channels. Overall, in developing BTC systems, we should be aware of the fact that there are strict delay and high data rate requirements for both downlink and uplink connections, with a high duty cycle, which calls for high energy efficiency. These are essential to ensure OoPE in wireless networks with "brain-in-the-loop". Moreover, brain implants must address biocompatibility issues, high channel density and data transfer, and communication security. In this context, we describe intelligent reflecting surfaces as a promising approach. In the next section, we will move toward our second path by describing the contributions that wireless communications can bring to neurosciences.

IV. WIRELESS NETWORKS FOR NEUROSCIENCES

Up to now, we have discussed different ways in which neurosciences would become part of the future wireless communication systems through BTC, possibly already with the upcoming 6G. In this section, we will describe how wireless communication theory and specific communication systems (in particular, beyond 5G systems) can support future research and technological development in neurosciences considering the development of BTC as described in the previous section.

A. BTC Performance for Neurosciences

While the plurality of applications are waiting for the real development of wireless BMIs, we must conduct an initial assessment of the existing metrics, or new metrics, that allow understanding of what is required in the definition of wireless communication systems for Brains-to-Wireless infrastructure connections. This initial analysis is made based on the recent breakthroughs in the 5G and Beyond 5G research, which is the cornerstone of 6G (supporting our argument that BTC applications could be included in its standardization process), as well as recent engineering advancements in neural interfaces, which are the central elements of BMIs. In this section, we will indicate the requirements that 6G, or any future wireless system to support BTC, needs to meet to allow for new neurosciencesbased applications. The key vision is to achieve fine granularity of brain functions from both sensing and actuation capabilities by integration with a specific wireless communication technology. Then, the performance expected of 6G or any other technology is drawn upon the ability of delivering enough performance that maintains the functioning of future BMIs for a long time with security and safety for users.

1) Data Rate: A naive estimation of the total brain recording demand is about 100 Gbits/s, which is not supported by existing and near-deployment 5G infrastructures. However, in the context of individual connections, this is a considerable demand for future technologies, which are currently not being considered because of the lack of popularity of BMIs. This estimation was also naively performed because it does not consider the current and future technology for BMI, which surely can increase this number as more and more techniques are capable of obtaining not only electrophysiological neuron signals but also signals from other cell types in the brain, and lastly, other types of information, such as biomarkers. On top of that, this naive estimation is also based on the standard sampling rate of neural signal acquisition (1 kHz), which varies between technologies and recording strategies. The needs for an increased data rate must deal with all the aforementioned information, even though it requires more investigation into the real data rate requirements of BMI. By looking at increased requirements for spectrum resources, one must keep in mind that BMI is one of the multiple applications that future wireless systems will likely accommodate. Together with multimedia, gaming, e-health applications, and more, BMI can increase

the burden on future network generations for more data rate requirements than previously expected.

2) Reliability: BMIs as a technology can open a wide variety of applications sensitive to network disruption. For example, remote treatment of epileptic patients will require a constant usage of the BMI for detecting random seizure events as well as treating the disease by using current stimulation techniques also driven by the BMI solution. In this case, the implications of network disruption are above from conventional application delays or stream interruptions that are commonly found in conventional networks. In this context, the disease control mechanisms that are based on a BMI solution could be disrupted in a way that either can start unpleasant symptoms in patients or not support advance signal processing techniques for temporal variant data that support diagnosis systems. Based on the assumption of BMIs actively being used in small cells, this means not only that high frequencies must be managed to provide reliable connections that are not interfered by obstacles and environmental molecular effects, such as water vapor. These phenomena are known as the major challenges in small cells for present beyond-5G technologies, where intelligent reflecting surfaces are being currently the best choice to provide highly reliable connections. However, this technology is far away from being mature to guarantee high levels of network reliability based on the primary focus on physical mechanisms of beam-steering of high frequencies as opposed to the study of network resilience, which must be the next step of the research in this topic. The importance of network reliability brings the focus again to solutions that maintain a constant data rate in certain applications, where conventional network solutions must be upgraded in future technologies, likely already in 6G.

3) Energy Management: Wireless BMIs based on brain implants will most likely operate on different wireless media than the wireless infrastructure. For example, while RF is an option for wireless BMIs, high frequencies are unlikely to be used because of signal absorption and scattering resulting from the high water molecule profile of the tissue and skull. However, the two better fitting options are magnetic induction and the ultrasound system. Their differences are highlighted by their performance profile; magnetic induction is better for the implant data rate, whereas the ultrasound system enables deepness of implantation. The major challenge is that future technologies including 6G will most likely operate around the sub-Terahertz bands, which means that constant frequency conversion is required in order to provide integration of wireless BMIs into the wireless infrastructure. Because frequency homogenization is not an option, it can be easily foreseen that BMIs must have short-term memory strategies that support the frequency translations without loss of data. The issue here is then that both the frequency conversion techniques and memory are energy expensive, which adds the concerns for the constant usage of BMIs for chronic patients, or other applications, such as streaming or gaming. Energy management solutions must emerge not only at the device level, but also at the network level, which can work together by using advanced protocols or virtual infrastructures that enable efficient and data lossless connections with BMIs.

4) Latency: Today's communication infrastructure is guided by techniques that provide massive ultrareliable and low-latency communication. This shall not change for communication with BMIs. The importance of these strategies is directly linked with the future of BMIs and their success, because the main goal is to allow constant daily usage for patients and other users. The radical societal change based on BMI will only happen when we are capable of using this technology integrated into our daily activities, either to support or enhance them. 6G and future technologies involve the idea of massive sensing, which fits into the future BMI technology that envisions hundreds or thousands of nano-scale devices that interface with neuronal cells. The information on that scale, i.e., the LFP, enables rich cellular information that is now used to make precise predictions of disease states and trigger events. In addition to that, the idea of massive stimulation can also be implemented, where these several devices will act on the neural tissue to stimulate whole cell populations or parts of them. Here, latency is crucial in order to operate these functions remotely while maintaining the safety and security of each user. At the same time, BMI has to be perfectly modeled and tackled in a way that allows scalability. Scalable BMIs are practically nonexistent, and 6G might as well be the technology needed to open these doors.

B. Wireless-Based Brain-Machine Interfaces

A more direct application of the future generation of wireless technology that neurosciences would benefit from would be a new generation of BMIs. BMIs have been used to alleviate motor deficits but also as a tool to characterize neural correlates of behavior [46]. Here, tethered neural recording systems are a major limitation because they hinder natural and social behavioral interactions. Most notably, in the late 2000s, novel wireless recording technologies have become available, which can simultaneously sample several hundreds of neurons from different brain regions (see [50] for a review of recording technologies).

Schwarz *et al.* [83] developed a bidirectional wireless system capable of implementing part of the signal processing pipeline at the headstage and transceivers attached to an animal's head. Four transceivers were used; each transceiver was connected to 128 recording channels sampled at 31.25 kHz per channel, consuming 2 mW per channel, with a total of 48 Mbps aggregate rate of data acquisition, and an optimal operating range of 3 m. The device was reported to be able to continuously operate for over 30 h. With the device implanted in a monkey, authors were able to record 494 neurons from four different brain regions. The animal successfully performed established BMI tasks wirelessly [84], which confirmed the suitability for studying natural, social interactions and complex movement behaviors.

More recently, the first wireless invasive BMI has been demonstrated in humans [85]. The interface had 192 electrodes, with 20 kSamples/s per electrode (12 bits per sample). Prior to wireless transmission, Manchester encoding was used to reduce error and improve reliability. Each data frame contained one 50 μ s 12-bit sample from all electrodes and was

transmitted at 3.3 GHz or 3.5 GHz. Recordings were carried out uninterruptedly for a 24 h period, and subjects successfully performed a computer cursor BMI task.

One major problem is common to invasive recording systems: implants inevitably cause lesions in the brain tissue, resulting in inflammation and limiting the sampling of deeper brain regions. To alleviate this problem, wireless sub-millimeter scale devices are being developed that can both record and stimulate neural activity. Ghanbari *et al.* [86] describe an ultrasonically powered neural recording implant, with simultaneous power-up and communication, that can achieve an over 35 kbps/mote equivalent uplink data rate. Thus, in principle, this device could operate as part of a wireless BMI.

In terms of noninvasive wireless BMIs, EEG is arguably the most common recording strategy. By avoiding surgical procedures, EEG has been widely used in human BMIs. However, EEG signals reflect the activity of millions of neurons from the surface of the brain, thus hindering decoding performance, which leads to a limited set of motor commands that can be extracted in noninvasive BMIs. Common commercial wireless devices range from 8 to 64 channels, with sampling frequencies of up to 1 kHz and a few dozen meters of transmission [87]. Nevertheless, modern hardware and computational intelligence methods, such as flexible electronics and deep learning, have been shown to boost wireless EEG-based BMI performance, reaching up to 122 bits per minute of information transfer rate [88].

C. Internet of Bio-Nano Things

Another key promising application is the IoBNT [89]. The IoBNT can aid the diversity of BMIs and their types by interacting with the brain using molecules, peptides, and molecular structures in general [90]. As it stands, no molecules are used to convey synthetic information of any type, which means that a whole biodiversity of information is being underutilized as opposed to enhanced means of communication between implantable devices and brain tissue. The research area of molecular communications promotes the usage of molecules as carriers for interactions between implantable-implantable and between implantable-biological systems [91]. Increased biocompatibility is thus reached when understanding and using molecules that are currently being used in biological systems, now with the purpose of controllable biological communication [92]. This infrastructure is envisioned to bridge to the Internet by means of synthetic biology and advanced nanotechnology, where electromagnetic-molecular signal translation is performed toward remote digital control of the internal cellular process of either Eukaryotic and prokaryotic cells.

The diversity of molecules inside a human body is assumed to be huge, and therefore, a means of translating molecular information between tissues is considered to be of great importance even with the limited investigation by the scientific community so far. The idea for the future technology is that there are internal synthetic cells capable of converging molecular information from different types of tissues and vice versa in order to support the idea of biomolecular intrabody networks [93]. Therefore, implantables or even bionanomachines located in different tissues can communicate with each other without the need of predetermined molecular coherence, which can empower flexibility and performance of these systems. At the edge of these networks, the molecular information is translated into electromagnetic information by biocyber interfaces that are also capable of translating the opposite case [94].

Inside the brain, implantables of bionanomachine devices have the main purpose of influencing the brain activity by manipulating the ionic channels that are understood to be a major part of the information propagation in the brain. There are a variety of molecules on the micro scales of the brain, including calcium, potassium, and sodium. Neurotransmitters and gliotransmitters are ions that regulate the information propagation inside the synaptic channel between neurons. These molecules have been studied and analyzed for many decades and are controlled to treat many neurodegenerative diseases. Brain-machine interfaces for molecular interactions have a huge impact in the future of the neurodegenerative diseases. The levels of control that can be reached by digital systems can be tremendously beneficial to chaotic systems, such as biological systems in the brain [95]. The main challenge is that the major biological properties of the brain have been well understood before considering them as control variables; the understanding of these properties is a timeconsuming effort that has to focus on neuroscientific efforts undertaken through many decades.

However, there are works that demonstrate the idea of IoBNT prevailing through existing disease challenges together with biotechnology as well as future oncology efforts. The EU-H2020-FET Gladiator project deploys a hybrid neural interface that is implanted into the brains of patients with glioblastoma, with the main goals of utilizing the modulation of drug propagation in the brain that maximizes drug efficacy while minimizing its side effects [96]. For that, wireless external signals control these hybrid neural interfaces to produce molecules that contain multiple drug molecules inside them, called exosomes. These exosomes are drug carriers that ultimately dictate how and when the tumors in the brain are being treated by this novel cancel therapy. The novel paradigm of molecular communication is used to characterize the data rate and capacity of exososome-based communication systems between the hybrid interface and the brain cancer. Here, the channel is understood to be the extracellular brain space in which the exosomes can propagate through a biased random motion. The many brain cells create tight spaces in which the exosomes propagate and where there is enough brain fluid to drive movement, called brain parechyma. The research is now focusing on the development of both theoretical and in-vitro models that demonstrate the above-mentioned system, which can radically change the existing state-of-the-art of cancer treatment methods.

D. Complex and Chaotic Communications to Quantify Neural Activity

Tools of communications and information theory can also provide interesting analytical approaches to assess the behavior and fundamental limits of neural communications, which, by nature, exhibit chaotic properties. Neurons, themselves, have a complex morphology that allows the connection to many other neurons. They have several morphological types, but also varied electrical activity profiles. A single neuron can be connected to thousands of other neurons and non-neuron cells through millions of synapses. Once these neurons comprise a large complex network of cells, its nonlinear effects result in complex or chaotic dynamics. The individual elements of the network contribute cumulatively to these high-order dynamics with their simple, yet diverse actions, which can be deterministic or stochastic themselves. Analyzing neural networks with modeling approaches based on complexity and/or chaos can provide rich dynamics that are governed by deterministic processes (i.e., action potentials) in order to obtain nontrivial patterns and behavior. This is essential to not only understand and characterize neural networks but at the same time provide communication systems that can interact with them.

The analysis of neural signals using chaos-based principles started with EEG signals [97]. It is now accepted, from measures of entropy, that the brain indeed presents itself, at least in certain modes of operation, as a chaotic-behaving system. For example, studies of diseases have been further analyzed using chaos-based principles, like schizophrenia, which is suggested to be a decay of complexity, either on the side of randomness (infinite entropy) or order (zero entropy) [98]. Another example is the effect of drugs, like LSD, that are connected to periodic neural oscillations, thus losing chaoticity and decreasing complexity [99].

Even though the effort to discover organization in nature had its origins in randomness, it was realized that measures of randomness do not capture the property of organization [100]. This is a very important issue in neurosciences, as the knowledge of organization and activity needs to be bridged in order to provide reliable communications with the brain [101], [102]. In signal processing, one can find measures that capture a system's complexity-organization, structure, memory, symmetry, and pattern. Complexity measures would allow us to quantify the hidden micro-level relationships between system parts that result in the system properties obtainable at the macro level, which would, for example, quantify the relationship between cellular to network levels [103]. The authors of [104], [105] argue that a complex system lives in between a random and a completely regular system, leading to the conclusion that a lot of the complexity metrics (e.g., Kolmogorov complexity in algorithmic information theory, dimensional complexity in neurobiology) do not measure "true complexity" because they do not attain small values for both random and regular systems. Random systems have no structure at any level, which results in high entropy and low complexity. On the other hand, regular systems exhibit low entropy and low complexity because of the repetition of structures at multiple levels. Therefore, it is obvious that complexity and entropy are two distinct quantities. As highlighted in [106], entropy captures the disorder and inhomogeneity rather than the correlation and structure of a system. Therefore, the measure of randomness, which has previously been applied to neurosciences to quantify neural activity, might be re-evaluated

in terms of capturing the true complexity of neuron activity inside the brain. It may also be used on multiple scales—for example, by linking spiking activity in subcortical regions to the whole brain. Even though we believe that we can recognize complexity when we see it, complexity is an attribute that is often without any conceptual clarity or quantification per se; however, because of the many unknowns in neurosciences, it should be further explored. For this purpose, the brain can also be studied as a complex communication network associated with structure and function, and evaluated with information-theory-inspired metrics and distributed communication system performance indicators. This knowledge translation from complex networks to a brain network can shed light on organization and structures of the brain that are currently unknown.

E. Lessons Learned

In this section, we indicated how the future generation of wireless systems, potentially 6G, would benefit the neurosciences community. We have shown the main requirements that, from our view, future technology would need to meet to incorporate BTC-based applications, BMI in particular. We highlight that high data rates, reliability, latency, and energy management should be the focus of research in wireless-based BMIs. In this context, considering the particularities of neuronal activity, we brought the discussion to a highly promising research path: IoBNT, which incorporates molecular-level communications. Finally, we have explored complexity sciences and chaos theory as theoretical tools that could be helpful to measure and analyze brain activity, and help the design of more effective dedicated communication systems.

V. CASE STUDY: BRAIN-CONTROLLED VEHICLES

In this section, we present the state-of-the-art of one particular application called brain-controlled vehicles (BCVs) and how we foresee its development based on 6G (or other future wireless technology). BCVs are an interesting, nonmedical application that involves both neurosciences for wireless networks and wireless networks for neurosciences. The automotive industry is one of the world's largest industries, highly competitive and exposed to novel technologies, and therefore, BCVs could be a disruptive factor in promoting the development of BTC technology. Clearly, advanced BCVs cannot rely on noninvasive technology, but rather on the novel neuroscientific and wireless technologies that we have discussed thus far. Therefore, the goal of the case study is to show one promising application in a highly relevant industry and emphasize how modern BMIs (based on invasive signals) and modern wireless communications could expand it.

A. State-of-the-Art

The field of BCVs with EEG-based BMI has experienced a steady growth since 2010, usually focusing on applications to support disabled patients. In addition to the already discussed challenges of BMI in relation to developing effective algorithms for feature extraction and classification, current



Fig. 5. Future direction of a BCV application, depicting a rescue mission. Users are seated in BTC-enabled chairs with sensors, wearables, and XR devices connected to the 6G network, and have seamless access to all relevant information for the mission. Neural spikes and molecular signals are registered and integrated into the IoBNT, supported by the wireless network infrastructure. Simultaneously, users also receive sensory feedback directly by the same invasive implants. Distributed AI algorithms ensure low-latency communication between users and vehicles as well as between brains. Ambient information, together with users' action intentions and emotional state, creates a broad structure of awareness which, altogether, contributes to a successful operation.

BMI hardware has well-known limitations concerning the communication range and speed.

Despite these challenges, several studies have demonstrated the feasibility of BCVs. We can mention, for instance, control of a vehicle in four main directions [107], [108], methods for obstacle avoidance [109], [110], and hand brake assistance in emergency situations [111], [112], using diverse platforms such as vehicle simulators, virtual reality vehicles, vehicles in video games, quadcopters, drones, helicopters, and fixed-wings aircrafts. A general simulator-based procedure for training a participant is shown in Fig. 5.

In a seminal work, Haufe *et al.* [113] implemented an assistant brake system in emergency cases for BCV applications based on EEG and electromyography (EMG) signals, which was tested on a simulated vehicle. The algorithm identifies brain activity patterns related to emergency braking intentions in a simulated graphical racing car task. In a similar vein, Kim *et al.* [114] attempted to detect the driver's emergency braking intention in different situations for a simulated vehicle based on EEG and EMG signals. This method was further improved in [115], including a real-world experimental task. In Göhring *et al.* [116], a semi-autonomous vehicle was implemented with different external sensors and a camera, and then controlled using EEG-based brain activity patterns. To control the vehicle, two different scenarios, obstacle avoidance and braking and steering, were used.

In a series of studies by Bi *et al.* [111], [117]–[121], different approaches to identify and predict the driver's intention for moving forward, turning left and right, as well as emergency braking, were studied. The development of AI-based learning methods is leading to improvements in those tasks, as reported in [108], [122]–[128]. Similar research has also been carried out to study BCVs for aerial vehicles, as in [129]–[131]. A major challenge lies in separating those features of brain signals that relate to vehicle control from those that are not. The second task to enhance the results is to develop or modify the existing classifiers into a highly accurate multiclassifier, such as a deep belief learning algorithm. The third limitation is the limited number of participants for training and testing of the algorithms.

Arguably, the major limitation on BCVs is the availability of more informative neural signals than those that EEG or EMG can provide. The use of noninvasive brain signals has led to significant contributions, in particular, better braking systems, but a comprehensive control of a vehicle requires fine-tuned actions. State-of-the-art EEG-based BMIs boosted by deep-learning methods have proven to be feasible for multidirectional 3D robotic arm control, with success rates of about 60% [132]. Even the more invasive methods, such as ECoG, hardly surpass the 80% success rate in 2D cursor control tasks [133]. Although these are impressive results with a wide range of practical applications, BCVs require stable, high success rates. Furthermore, noninvasive methods have access to cortical information, the outermost layer of the brain, whereas other deeper brain regions, like the cerebellum or basal ganglia, may convey information that is fundamental for a successful vehicle operation.

For a comparison, Gopal *et al.* [134] report a tethered 96electrode BMI with a performance of 6.5 bps, which would correspond to typing 15 words/min with a basic alphanumeric keyboard, whereas a standard EEG-based BMI has a performance of ~1.0 bps, or 3 words/min [135]. Considering the BMI performance scales with the number of neurons recorded, modern prototype interfaces with thousands of electrodes, such as the one developed by Neuralink [8], indicate a promising future for BCV. As we discussed in Section III, the progress in direct brain implants that communicate wirelessly, supported by 6G or other technologies, will enhance the repertoire of brain commands to the vehicle and also facilitate bidirectional communication, in which signals from the vehicle and its surroundings may be delivered directly to the driver's brain. In addition, future wireless networks would need to accommodate BTC protocols and AI algorithms that can integrate all of the required processing stages for a robust, pleasant, and secure driving experience.

B. Future Direction: Neurosciences–Wireless Networks Converged Applications

Although successfully tested under different conditions, BCVs as designed today are not a scalable solution as they would require a wireless connection to support BTC with high coverage, availability, speed, and low latency to provide reliability and safety for the end-users. As discussed above, despite the great development of wireless communications (remarkably 5G), the existing solutions would not work today because of the stringent requirements of BTC (see Section III). However, if the path indicated in this paper was realized, a scalable BCVs would become feasible by using a new generation of wireless-connected BMI with 6Gconnected high-density implants supported by IRSs to enable BTC. This would also be associated with the possibility of acquiring and processing more biosignals via IoTBNT, linked with Intelligent Sensor Networks (ISNs) that could sample the environment in which the BCV is moving. Furthermore, AI-enabled distributed cloud algorithms could support sophisticated signal processing in the fraction of the second scale required by a safe driving experience. The performance limits of such communicative brain devices could be derived from information- and communication-theoretical tools applied for chaotic and spiking systems, while new chaos-based waveforms for communication might also be developed.

As a rough example, we could imagine the following future scenario in 10-15 years from now. A major fire outbreak triggers an emergency response central, and multiple BCV, both aerial and terrestrial, are sent to the dangerous site. Modern wireless networks, with distributed AI algorithms, support pervasive coverage, ultrahigh throughput, and low-latency communication. Remote operators are seated in BTC-enabled chairs with sensors, wearables, and XR devices connected to the 6G network, and have seamless access to all relevant information for the mission. The users can communicate directly via brain-to-brain communication. Invasive implants, together with the IRS and IoBNT infrastructure, establish a bidirectional link with the users' brains to operate the semiautonomous vehicles. Ambient information, users' action intentions, and emotional state create a broad structure of awareness, which, altogether, contributes to a successful operation.

Clearly, this scenario could be extended and rethought, but it illustrates a potential future that we believe is technologically feasible given the state-of-the-art in wireless communications and neurosciences, as well as in biosignal processing and computer sciences. The main lesson to be learned is that the convergence of those fields is more likely to happen if stimulated by the definition of the future generation of wireless systems. In particular, having BTC as part of 6G research agenda would provide clear guidance of how those kinds of potential futures could become a reality for many applications related to future 6G-connected BMIs, as the BCV example indicated.

VI. SECURITY, PRIVACY, AND ETHICAL ASPECTS

Current BMIs are applied mostly in the medical and therapeutic context, in which rigid protocols regarding experimental design and the usage of data are enforced. A substantial body of literature has proven that it is possible to decode and manipulate brain activity, thus granting access to one's feelings, emotions, and intentions in an unprecedented way. Clearly, one of the main challenges for the development of BTC systems relate to security, privacy, and ethical aspects in BMIs [136]–[138].

First and foremost, despite the astonishing progress in recording technology, it is still unclear if a long-term neural implant can be placed without damaging neural tissue (see Section II-C). This is definitely one of the crucial steps toward BTCs, because invasive recordings offer access to biosignals unavailable to noninvasive methods. Thus, the futures of BTC and neural recording technology are intertwined. Novel works offer promising initiatives, from nanorobotics to molecular communication [93].

Then, the capacity to read from and write to an individual's brain opens access points to memories and may even be used to change one's behavior. Even if implants and communication networks are safe, BTCs mean that artificial devices will constantly interact with individuals, possibly shaping their agency and affecting social behavior. Moreover, human reason becomes distributed and the sense of responsibility is dramatically impaired: if a harmful action results from the operation of a BMI, who should be accounted for the damage? The ethical controversy underlying self-driving vehicles gives a indication that ethics may play as important a role for technological development as technical challenges [137].

To ensure that novel technologies that exploit BMIs abide by international standards and human rights, some of the most representative researchers in the field have claimed for guidelines to be established [139]–[141]. Overall, four main concerns are highlighted: privacy and consent; agency and identity; augmentation; and bias. In this context, neural information and neural applications should be strictly regulated, decentralized, and subject to transparent social scrutiny and user consent, especially if military purposes are considered.

From a different but related perspective, established communication networks approaches have addressed security, privacy, and ethical aspects of their usage, regardless of the underlying technologies. The advent of BTCs and novel BMIs complicate this nontrivial debate, considering that novel applications emerge in parallel with the standardization protocols. More recently, even though 5G networks are yet to be fully deployed, there has been an intense security and privacy concern [142], [143]. In comparison with 5G, 6G, or other potential future wireless networks will boost real-time responsive systems, capable of AI-based autonomous decisions, interfacing brains and machines. This draws attention to specific vulnerabilities related to authentication, access control, malicious behavior, encryption, and data transmission, which are closely linked to the novel technologies that support future technologies [144].

In addition, the advent of two-way BTC systems is accompanied by novel vulnerabilities, such as neuronal cyberattacks to brains [145], which are still largely ignored in the BMI literature. Until now, the possibility of influence on human behavior by directly modulating the brain has always been limited to clinical environments, but here we envision that future BMIs and BTC systems must take into account the hypothesis of undesired signals, not necessarily malicious, that affect brain implants.

Several solutions have been proposed [146]–[149]. A central aspect is to restrict the centralized processing of neural information and ensure privacy. For that, blockchain-like mechanisms, differential privacy, and federated learning offer promising alternatives. Concurrently, international governments and regulatory bodies should agree on guidelines for the usage of neurotechnology, similar to what has been done for nuclear energy or gene editing applications.

VII. CONCLUDING REMARKS AND LESSONS LEARNED

This tutorial paper provided an in-depth overview of an interdisciplinary research field at the intersection of neurosciences and wireless communications, as well as signal processing, control theory, and computer sciences. We argue here that an organic encounter between these two research areas has the full potential to take place in future developments of wireless networks, in particular 6G, which will support BTC considering not only its strict requirements but also the particularities of neural signals and brain communication. By revisiting the literature, we have classified the expected benefits of this joint research into two groups, from where we can state the following key research directions.

- Neurosciences for wireless networks focuses on how developments in neurosciences will enable new application in the next generation of wireless systems based on BTC, in contrast to HTC and MTC. The key challenges relate to latency, high data rate, and high energy efficiency requirements, which are essential to support QoPE in BTC. Studies on the nature of the neural code, and how to match neural dynamics to behavior, are imperative to shed light on communication protocols. In a complementary effort, future works on materials and biomedical engineering should move toward brain implants with a higher channel density whilst ensuring biocompatibility and security.
- Wireless networks for neurosciences focus on how future wireless technologies could support new research and development in neurosciences, potentially including novel wireless-enabled BMIs, and IoBNT, as well

as information- and communication-theoretic ways of evaluating brain communications based on their chaotic nature. In this context, the core future developments lie on the strict reliability and latency requirements, considering critical applications, such as BCVs. Furthermore, novel research fields, such as IoBNT, which incorporate molecular-level communications, will increase the demand for high data rates. Finally, having communicating brains in wireless networks bring forth serious ethical and security issues that have to be addressed prior to deployment of any BTC technology.

We illustrated the potential benefits of this proposed research agenda by analyzing a brain-controlled vehicle application.

We expect this contribution to serve as a key reference for researchers from both domains to start building joint activities that are necessary to realize the vision indicated here. The proposed discussions shall point toward a direction full of potential, from basic research to product development, but this can only be realized as a truly interdisciplinary task, similar to the path taken by neuromorphic computing [2]. In particular, a fully integrated smart city society immersed in an ubiquitous wireless computation environment will certainly find its great dilemmas in security, privacy, and ethics, and these topics must underlie any endeavor. In summary, although we are aware that the relation between neurosciences and wireless communications is still not fully established, this tutorial clearly indicates its feasibility if enough efforts are dedicated toward this goal. Our view is that this task needs to be put forth now during the discussions of novel wireless technologies, including 6G, so that BTC enters the agenda of the standardization bodies, indicating the research path for the coming ten years.

ACKNOWLEDGMENT

The authors would like to thank Dr. Hanna Niemelä for valuable comments and for proofreading this paper.

REFERENCES

- M. Chen, U. Challita, W. Saad, C. Yin, and M. Debbah, "Artificial neural networks-based machine learning for wireless networks: A tutorial," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 4, pp. 3039–3071, 4th Quart., 2019.
- [2] D. Strukov, G. Indiveri, J. Grollier, and S. Fusi, "Building braininspired computing," *Nat. Commun.*, vol. 10, no. 1, p. 4838, 2019.
- [3] K. Bonawitz et al., "Towards federated learning at scale: System design," 2019. [Online]. Available: arXiv:1902.01046.
- [4] T. Li, A. K. Sahu, A. Talwalkar, and V. Smith, "Federated learning: Challenges, methods, and future directions," *IEEE Signal Process. Mag.*, vol. 37, no. 3, pp. 50–60, May 2020.
- [5] M. Chen, Z. Yang, W. Saad, C. Yin, S. Cui, and H. V. Poor, "A joint learning and communications framework for federated learning over wireless networks," *IEEE Trans. Wireless Commun.*, vol. 20, no. 1, pp. 269–283, Jan. 2021. [Online]. Available: https://arxiv.org/pdf/1909.07972.pdf
- [6] M. Chen, W. Saad, H. V. Poor, and S. Cui, "Convergence time minimization of federated learning over wireless networks," in *Proc. Int. Conf. Commun.*, Dublin, Ireland, Jun. 2020, pp. 1–6.
- [7] M. Chen, O. Semiari, W. Saad, X. Lin, and C. Yin, "Federated echo state learning for minimizing breaks in presence in wireless virtual reality networks," *IEEE Trans. Wireless Commun.*, vol. 19, no. 1, pp. 177–191, Jan. 2020.

- [8] E. Musk, "An integrated brain-machine interface platform with thousands of channels," *J. Med. Internet Res.*, vol. 21, no. 10, Oct. 2019, Art. no. e16194. [Online]. Available: http://www.jmir.org/2019/10/e16194/
- [9] M. Pais-Vieira, G. Chiuffa, M. Lebedev, A. Yadav, and M. A. L. Nicolelis, "Building an organic computing device with multiple interconnected brains," *Sci. Rep.*, vol. 5, pp. 1–15, Jul. 2015.
- [10] W. Saad, M. Bennis, and M. Chen, "A vision of 6G wireless systems: Applications, trends, technologies, and open research problems," *IEEE Netw.*, vol. 34, no. 3, pp. 134–142, May/Jun. 2020.
- [11] "White paper on RF enabling 6G published," 6G Channel, Oulu, Finland, White Paper, Aug. 2020. [Online]. Available: https://www.6gchannel.com/6g-white-papers/
- [12] C. Chaccour, R. Amer, B. Zhou, and W. Saad, "On the reliability of wireless virtual reality at terahertz (THz) frequencies," in *Proc.* 10th IFIP Int. Conf. New Technol. Mobility Security (NTMS) Mobility Wireless Netw. Track, Canary Islands, Spain, Jun. 2019, pp. 1–5.
- [13] M. Chen, W. Saad, and C. Yin, "Virtual reality over wireless networks: Quality-of-service model and learning-based resource management," *IEEE Trans. Commun.*, vol. 66, no. 11, pp. 5621–5635, Nov. 2018.
- [14] S. Dang, O. Amin, B. Shihada, and M.-S. Alouini, "What should 6G be?" *Nat. Electron.*, vol. 3, no. 1, pp. 20–29, 2020.
- [15] M. A. Hofman, "Design principles of the human brain: An evolutionary perspective," *Progr. Brain Res.*, vol. 195, pp. 373–390, Dec. 2012.
- [16] A. M. M. Sousa, K. A. Meyer, G. Santpere, F. O. Gulden, and N. Sestan, "Evolution of the human nervous system function, structure, and development," *Cell*, vol. 170, no. 2, pp. 226–247, 2017.
- [17] S. B. Laughlin and T. J. Sejnowski, "Communication in neuronal networks," *Science*, vol. 301, no. 5641, pp. 1870–1874, 2003.
- [18] B. Wang *et al.*, "Firing frequency maxima of fast-spiking neurons in human, monkey, and mouse neocortex," *Front. Cell. Neurosci.*, vol. 10, pp. 1–13, Oct. 2016.
- [19] S. Herculano-Houzel, "Scaling of brain metabolism with a fixed energy budget per neuron: Implications for neuronal activity, plasticity and evolution," *PLoS ONE*, vol. 6, no. 3, 2011, Art. no. e17514.
- [20] J. E. Niven and S. B. Laughlin, "Energy limitation as a selective pressure on the evolution of sensory systems," *J. Exp. Biol.*, vol. 211, no. 11, pp. 1792–1804, 2008.
- [21] R. von Bernhardi, L. E.-V. Bernhardi, and J. Eugenín, What Is Neural Plasticity?. Cham, Switzerland: Springer, 2017, pp. 1–15. [Online]. Available: https://doi.org/10.1007/978-3-319-62817-2_1
- [22] G. Buzsáki, C. A. Anastassiou, and C. Koch, "The origin of extracellular fields and currents—EEG, ECoG, LFP and spikes," *Nat. Rev. Neurosci.*, vol. 13, pp. 407–420, Jun. 2012. [Online]. Available: http://dx.doi.org/10.1038/nrn3241
- [23] E. R. Kandel, *Principles of Neural Science*. New York, NY, USA: McGraw-Hill Educ., Jun. 2000.
- [24] C. Rossant *et al.*, "Spike sorting for large, dense electrode arrays," *Nat. Neurosci.*, vol. 19, no. 4, pp. 634–641, 2016.
- [25] M. A. Lebedev and M. A. Nicolelis, "Brain-machine interfaces: From basic science to neuroprostheses and neurorehabilitation," *Physiol. Rev.*, vol. 97, no. 2, pp. 767–837, 2017.
- [26] V. Aggarwal, M. Mollazadeh, A. G. Davidson, M. H. Schieber, and N. V. Thakor, "State-based decoding of hand and finger kinematics using neuronal ensemble and LFP activity during dexterous reach-tograsp movements," *J. Neurophysiol.*, vol. 109, no. 12, pp. 3067–3081, 2013.
- [27] M. Steriade, Neuronal Substrates of Sleep and Epilepsy. Cambridge, U.K.: Cambridge Univ. Press, 2003.
- [28] A. H. Marblestone *et al.*, "Physical principles for scalable neural recording," *Front. Comput. Neurosci.*, vol. 7, p. 137, Oct. 2013.
- [29] C. Im and J.-M. Seo, "A review of electrodes for the electrical brain signal recording," *Biomed. Eng. Lett.*, vol. 6, no. 3, pp. 104–112, 2016.
- [30] B. J. Edelman *et al.*, "Noninvasive neuroimaging enhances continuous neural tracking for robotic device control," *Sci. Robot.*, vol. 4, no. 31, 2019, Art. no. eaaw6844. [Online]. Available: https://robotics.sciencemag.org/content/4/31/eaaw6844
- [31] C. Clément, Brain-Computer Interface Technologies. Cham, Switzerland: Springer, 2019.
- [32] J. del R. Milan and J. M. Carmena, "Invasive or noninvasive: Understanding brain-machine interface technology [conversations in BME]," *IEEE Eng. Med. Biol. Mag.*, vol. 29, no. 1, pp. 16–22, Jan./Feb. 2010.
- [33] S. Waldert, "Invasive vs. non-invasive neuronal signals for brainmachine interfaces: Will one prevail?" *Front. Neurosci.*, vol. 10, p. 295, Jun. 2016. [Online]. Available: https://www.frontiersin.org/ article/10.3389/fnins.2016.00295

- [34] R. K. Shepherd, Neurobionics: The Biomedical Engineering of Neural Prostheses. Hoboken, NJ, USA: Wiley, 2016.
- [35] N. A. Steinmetz, C. Koch, K. D. Harris, and M. Carandini, "Challenges and opportunities for large-scale electrophysiology with neuropixels probes," *Current Opinion Neurobiol.*, vol. 50, pp. 92–100, Jun. 2018. [Online]. Available: https://www.sciencedirect.com/science/article/pii/ S0959438817303161
- [36] S. M. Wellman *et al.*, "A materials roadmap to functional neural interface design," *Adv. Funct. Mater.*, vol. 28, no. 12, 2018, Art. no. 1701269.
- [37] M. A. M. Freire *et al.*, "Comprehensive analysis of tissue preservation and recording quality from chronic multielectrode implants," *PLoS ONE*, vol. 6, no. 11, pp. 1–15, Nov. 2011. [Online]. Available: https://doi.org/10.1371/journal.pone.0027554
- [38] P. O. Newton, "Spinal growth tethering: Indications and limits," Ann. Transl. Med., vol. 8, no. 2, p. 27, 2020.
- [39] S. M. Won, L. Cai, P. Gutruf, and J. A. Rogers, "Wireless and battery-free technologies for neuroengineering," *Nat. Biomed. Eng.*, to be published. [Online]. Available: https://doi.org/10.1038/s41551-021-00683-3
- [40] J. E. Ferguson and A. D. Redish, "Wireless communication with implanted medical devices using the conductive properties of the body," *Expert Rev. Med. Devices*, vol. 8, no. 4, pp. 427–433, 2011.
- [41] N. S. Agha, J. Komar, M. Yin, D. A. Borton, and A. Nurmikko, "A fully wireless platform for correlating behavior and neural data from an implanted, neural recording device: Demonstration in a freely moving swine model," in *Proc. 6th Int. IEEE/EMBS Conf. Neural Eng. (NER)*, San Diego, CA, USA, 2013, pp. 989–992.
- [42] D. A. Borton, M. Yin, J. Aceros, and A. Nurmikko, "An implantable wireless neural interface for recording cortical circuit dynamics in moving primates," *J. Neural Eng.*, vol. 10, no. 2, 2013, Art. no. 026010.
- [43] M. Yin *et al.*, "Wireless neurosensor for full-spectrum electrophysiology recordings during free behavior," *Neuron*, vol. 84, no. 6, pp. 1170–1182, 2014.
- [44] D. Seo *et al.*, "Wireless recording in the peripheral nervous system with ultrasonic neural dust," *Neuron*, vol. 91, no. 3, pp. 529–539, 2016.
- [45] R. Pfeifer and J. C. Bongard, *How the Body Shapes the Way We Think:* A New View of Intelligence (Bradford Books). Cambridge, MA, USA: MIT Press, 2006.
- [46] J. R. Wolpaw, N. Birbaumer, D. J. McFarland, G. Pfurtscheller, and T. M. Vaughan, "Brain–computer interfaces for communication and control," *Clin. Neurophysiol.*, vol. 113, no. 6, pp. 767–791, 2002. [Online]. Available: http://www.sciencedirect.com/science/article/pii/ S1388245702000573
- [47] A. L. Orsborn, H. G. Moorman, S. A. Overduin, M. M. Shanechi, D. F. Dimitrov, and J. M. Carmena, "Closed-loop decoder adaptation shapes neural plasticity for skillful neuroprosthetic control," *Neuron*, vol. 82, no. 6, pp. 1380–1393, 2014. [Online]. Available: http://dx.doi.org/10.1016/j.neuron.2014.04.048
- [48] A. R. C. Donati *et al.*, "Long-term training with a brain-machine interface-based gait protocol induces partial neurological recovery in paraplegic patients," *Sci. Rep.*, vol. 6, pp. 1–16, Aug. 2016.
- [49] N. Thakor, Building Brain Machine Interfaces—Neuroprosthetic Control with Electrocorticographic Signals, IEEE Lifesci. Newslett., Piscataway, NJ, USA, Apr. 2012.
- [50] G. Hong and C. M. Lieber, "Novel electrode technologies for neural recordings," *Nat. Rev. Neurosci.*, vol. 20, no. 6, pp. 330–345, 2019. [Online]. Available: http://dx.doi.org/10.1038/s41583-019-0140-6
- [51] R. Chen, A. Canales, and P. Anikeeva, "Neural recording and modulation technologies," *Nat. Rev. Mater.*, vol. 2, no. 2, pp. 1–16, 2017. [Online]. Available: http://dx.doi.org/10.1038/natrevmats.2016.93
- [52] M. M. Shanechi, "Brain-machine interface control algorithms," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 25, no. 10, pp. 1725–1734, Oct. 2017.
- [53] G. Buzsáki, "Neural syntax: Cell assemblies, synapsembles and readers," *Neuron*, vol. 68, no. 3, pp. 362–385, 2010.
- [54] G. B. Stanley, "Reading and writing the neural code," Nat. Neurosci., vol. 16, no. 3, pp. 259–263, 2013. [Online]. Available: http://dx.doi.org/10.1038/nn.3330
- [55] J. E. Lisman and O. Jensen, "The theta-gamma neural code," *Neuron*, vol. 77, no. 6, pp. 1002–1016, 2013. [Online]. Available: http://dx.doi.org/10.1016/j.neuron.2013.03.007
- [56] M. M. Churchland *et al.*, "Stimulus onset quenches neural variability: A widespread cortical phenomenon," *Nat. Neurosci.*, vol. 13, no. 3, pp. 369–378, 2010. [Online]. Available: http://dx.doi.org/10.1038/nn.2501

- [57] M. A. L. Nicolelis and M. A. Lebedev, "Principles of neural ensemble physiology underlying the operation of brain-machine interfaces," *Nat. Rev. Neurosci.*, vol. 10, no. 2, pp. 1530–1540, 2009.
- [58] G. Buzsáki, *Rhythms of the Brain*. New York, NY, USA: Oxford Univ. Press, 2006. [Online]. Available: http://www.oxfordscholarship. com/view/10.1093/acprof:oso/9780195301069.001.0001/acprof-9780195301069
- [59] M. A. Lebedev *et al.*, "Cortical ensemble adaptation to represent velocity of an artificial actuator controlled by a brain-machine interface," *J. Neurosci.*, vol. 25, no. 19, pp. 4681–4693, 2005.
- [60] A. C. Koralek, X. Jin, J. D. Long II, R. M. Costa, and J. M. Carmena, "Corticostriatal plasticity is necessary for learning intentional neuroprosthetic skills," *Nature*, vol. 483, no. 7389, pp. 331–335, 2012. [Online]. Available: http://dx.doi.org/10.1038/nature10845
- [61] P. Zioga, F. Pollick, M. Ma, P. Chapman, and K. Stefanov, "Enheduanna—A manifesto of falling' live brain-computer cinema performance: Performer and audience participation, cognition and emotional engagement using multi-brain BCI interaction," *Front. Neurosci.*, vol. 12, p. 191, Apr. 2018.
- [62] T. S. Rappaport *et al.*, "Wireless communications and applications above 100 GHz: Opportunities and challenges for 6G and beyond," *IEEE Access*, vol. 7, pp. 78729–78757, 2019.
- [63] S. Petkoski, A. Spiegler, T. Proix, and V. Jirsa, "Effects of multimodal distribution of delays in brain network dynamics," *BMC Neurosci.*, vol. 16, no. S1, p. P109, 2015.
- [64] S. Ali, W. Saad, and N. Rajatheva, "Fast uplink grant for machine type communications: Challenges and opportunities," *IEEE Commun. Mag.*, vol. 57, no. 3, pp. 97–103, Mar. 2019.
- [65] R. A. Ramadan and A. V. Vasilakos, "Brain computer interface: Control signals review," *Neurocomputing*, vol. 223, pp. 26–44, Feb. 2017.
- [66] D. Kahneman and A. Tversky, "Prospect theory: An analysis of decision under risk," *Econometrica*, vol. 47, pp. 263–291, Mar. 1979.
- [67] W. Saad, A. L. Glass, N. B. Mandayam, and H. V. Poor, "Toward a consumer-centric grid: A behavioral perspective," *Proc. IEEE*, vol. 104, no. 4, pp. 865–882, Apr. 2016.
- [68] C. F. Camerer, T.-H. Ho, and J.-K. Chong, "A cognitive hierarchy model of games," *Quart. J. Econ.*, vol. 119, pp. 861–898, Aug. 2004.
- [69] N. Abuzainab, W. Saad, C. S. Hong, and H. V. Poor, "Cognitive hierarchy theory for distributed resource allocation in the Internet of Things," *IEEE Trans. Wireless Commun.*, vol. 16, no. 12, pp. 7687–7702, Dec. 2017.
- [70] N. Rajatheva et al., "Scoring the terabit/s goal:Broadband connectivity in 6G," 2021. [Online]. Available: https://arxiv.org/abs/2008.07220
- [71] R. D. Yates, P. Ciblat, A. Yener, and M. Wigger, "Age-optimal constrained cache updating," in *Proc. IEEE Int. Symp. Inf. Theory (ISIT)*, Aachen, Germany, 2017, pp. 141–145.
- [72] B. Zhou and W. Saad, "Minimum age of information in the Internet of Things with non-uniform status packet sizes," *IEEE Trans. Wireless Commun.*, vol. 19, no. 3, pp. 1933–1947, Mar. 2020.
- [73] B. Zhou and W. Saad, "Joint status sampling and updating for minimizing age of information in the Internet of Things," *IEEE Trans. Commun.*, vol. 67, no. 11, pp. 7468–7482, Nov. 2019.
- [74] A. T. Z. Kasgari, W. Saad, and M. Debbah, "Human-in-the-loop wireless communications: Machine learning and brain-aware resource management," *IEEE Trans. Commun.*, vol. 67, no. 11, pp. 7727–7743, Nov. 2019.
- [75] A. O. Holcombe, "Seeing slow and seeing fast: Two limits on perception," *Trends Cogn. Sci.*, vol. 13, no. 5, pp. 216–221, 2009. [Online]. Available: https://www.sciencedirect.com/science/article/pii/ S1364661309000680
- [76] A. P. Georgopoulos, J. F. Kalaska, and J. T. Massey, "Spatial trajectories and reaction times of aimed movements: Effects of practice, uncertainty, and change in target location," *J. Neurophysiol.*, vol. 46, no. 4, pp. 725–743, 1981. [Online]. Available: https://doi.org/10.1152/jn.1981.46.4.725
- [77] Y. Su, S. Routhu, K. S. Moon, S. Q. Lee, W. Youm, and Y. Ozturk, "A wireless 32-channel implantable bidirectional brain machine interface," *Sensors*, vol. 16, no. 10, p. 1582, Oct. 2016.
- [78] B. Kibret, A. K. Teshome, and D. T. H. Lai, "Analysis of the human body as an antenna for wireless implant communication," *IEEE Trans. Antennas Propag.*, vol. 64, no. 4, pp. 1466–1476, Apr. 2016.
- [79] D. Seo, J. M. Carmena, J. M. Rabaey, E. Alon, and M. M. Maharbiz, "Neural dust: An ultrasonic, low power solution for chronic brainmachine interfaces," 2013. [Online]. Available: arXiv:1307.2196.
- [80] IEEE Standard for Safety Levels With Respect to Human Exposure to Electric, Magnetic, and Electromagnetic Fields, 0 Hz to 300 GHz, IEEE Standard Std C95.1-2019, pp. 1–312, 2019.

- [81] H. N. Schwerdt *et al.*, "A fully passive wireless microsystem for recording of neuropotentials using RF backscattering methods," *J. Microelectromech. Syst.*, vol. 20, no. 5, pp. 1119–1130, Oct. 2011.
- [82] A. S. de Sena *et al.*, "What role do intelligent reflecting surfaces play in multi-antenna non-orthogonal multiple access?" *IEEE Wireless Commun.*, vol. 27, no. 5, pp. 24–31, Oct. 2020.
- [83] D. A. Schwarz *et al.*, "Chronic, wireless recordings of large-scale brain activity in freely moving rhesus monkeys," *Nat. Methods*, vol. 11, no. 6, pp. 670–676, 2014.
- [84] S. Rajangam *et al.*, "Wireless cortical brain-machine interface for whole-body navigation in primates," *Sci. Rep.*, vol. 6, pp. 1–13, Mar. 2016. [Online]. Available: http://dx.doi.org/10.1038/srep22170
- [85] J. D. Simeral *et al.*, "Home use of a percutaneous wireless intracortical brain-computer interface by individuals with tetraplegia," *IEEE Trans. Biomed. Eng.*, vol. 68, no. 7, pp. 2313–2325, Jul. 2021. [Online]. Available: https://ieeexplore.ieee.org/document/9390339
- [86] M. M. Ghanbari *et al.*, "A sub-mm³ ultrasonic free-floating implant for multi-mote neural recording," *IEEE J. Solid-State Circuits*, vol. 54, no. 11, pp. 3017–3030, Nov. 2019.
- [87] C.-T. Lin et al., "Review of wireless and wearable electroencephalogram systems and brain-computer interfaces—A mini-review," *Gerontology*, vol. 56, no. 1, pp. 112–119, 2010.
- [88] M. Mahmood *et al.*, "Fully portable and wireless universal brainmachine interfaces enabled by flexible scalp electronics and deep learning algorithm," *Nat. Mach. Intell.*, vol. 1, no. 9, pp. 412–422, 2019. [Online]. Available: http://dx.doi.org/10.1038/s42256-019-0091-7
- [89] I. F. Akyildiz, M. Pierobon, S. Balasubramaniam, and Y. Koucheryavy, "The Internet of bio-nano things," *IEEE Commun. Mag.*, vol. 53, no. 3, pp. 32–40, Mar. 2015.
- [90] S. Balasubramaniam, S. A. Wirdatmadja, M. T. Barros, Y. Koucheryavy, M. Stachowiak, and J. M. Jornet, "Wireless communications for optogenetics-based brain stimulation: Present technology and future challenges," *IEEE Commun. Mag.*, vol. 56, no. 7, pp. 218–224, Jul. 2018.
- [91] I. F. Akyildiz, F. Brunetti, and C. Blázquez, "Nanonetworks: A new communication paradigm," *Comput. Netw.*, vol. 52, no. 12, pp. 2260–2279, 2008.
- [92] M. T. Barros and S. Dey, "Feed-forward and feedback control in astrocytes for Ca²⁺2+-based molecular communications nanonetworks," *IEEE/ACM Trans. Comput. Biol. Bioinformat.*, vol. 17, no. 4, pp. 1174–1186, Jul./Aug. 2020.
- [93] M. T. Barros, S. Balasubramaniam, and B. Jennings, "Comparative end-to-end analysis of Ca²⁺-signaling-based molecular communication in biological tissues," *IEEE Trans. Commun.*, vol. 63, no. 12, pp. 5128–5142, Dec. 2015.
- [94] U. A. K. Chude-Okonkwo, R. Malekian, and B. T. Maharaj, "Biologically inspired bio-cyber interface architecture and model for Internet of bio-nanothings applications," *IEEE Trans. Commun.*, vol. 64, no. 8, pp. 3444–3455, Aug. 2016.
- [95] M. T. Barros, "Ca²⁺-signaling-based molecular communication systems: Design and future research directions," *Nano Commun. Netw.*, vol. 11, pp. 103–113, Mar. 2017.
- [96] M. Veletić, M. T. Barros, I. Balasingham, and S. Balasubramaniam, "A molecular communication model of exosome-mediated brain drug delivery," in *Proc. 6th Annu. ACM Int. Conf. Nanoscale Comput. Commun.*, 2019, pp. 1–7.
- [97] J. Fell, J. Röschke, and P. Beckmann, "Deterministic chaos and the first positive Lyapunov exponent: A nonlinear analysis of the human electroencephalogram during sleep," *Biol. Cybern.*, vol. 69, no. 2, pp. 139–146, 1993.
- [98] A. C. Yang *et al.*, "Decreased resting-state brain activity complexity in schizophrenia characterized by both increased regularity and randomness," *Hum. Brain Map.*, vol. 36, no. 6, pp. 2174–2186, 2015.
- [99] J. Dubois and R. VanRullen, "Visual trails: Do the doors of perception open periodically?" PLoS Biol., vol. 9, no. 5, 2011, Art. no. e1001056.
- [100] D. P. Feldman, C. S. McTague, and J. P. Crutchfield, "The organization of intrinsic computation: Complexity-entropy diagrams and the diversity of natural information processing," *Chaos Interdiscipl. J. Nonlinear Sci.*, vol. 18, no. 4, 2008, Art. no. 043106.
- [101] M. T. Barros, H. Siljak, A. Ekky, and N. Marchetti, "A topology inference method of cortical neuron networks based on network tomography and the Internet of bio-nano things," *IEEE Netw. Lett.*, vol. 1, no. 4, pp. 142–145, Dec. 2019.
- [102] M. T. Barros, H. Siljak, P. Mullen, C. Papadias, J. Hyttinen, and N. Marchetti, "Objective multi-variable classification and inference of biological neuronal networks," 2020. [Online]. Available: arXiv:2003.12670.

- [103] M. T. Barros, W. Silva, and C. D. M. Regis, "The multi-scale impact of the Alzheimer's disease on the topology diversity of astrocytes molecular communications nanonetworks," *IEEE Access*, vol. 6, pp. 78904–78917, 2018.
- [104] G. Tononi, G. M. Edelman, and O. Sporns, "Complexity and coherency: Integrating information in the brain," *Trends Cogn. Sci.*, vol. 2, no. 12, pp. 474–484, 1998.
- [105] W. Bialek, I. Nemenman, and N. Tishby, "Complexity through nonextensivity," *Physica A, Stat. Mech. Appl.*, vol. 302, nos. 1–4, pp. 89–99, 2001.
- [106] D. P. Feldman and J. P. Crutchfield, "Structural information in twodimensional patterns: Entropy convergence and excess entropy," *Stat. Phys. Plasmas Fluids Relat. Interdiscip. Top.*, vol. 67, no. 5, pp. 1–9, 2003.
- [107] Y. Lu and L. Bi, "Combined lateral and longitudinal control of EEG signals-based brain-controlled vehicles," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 27, no. 9, pp. 1732–1742, Sep. 2019.
- [108] A. Hekmatmanesh, H. Wu, M. Li, A. M. Nasrabadi, and H. Handroos, "Optimized mother wavelet in a combination of wavelet packet with detrended fluctuation analysis for controlling a remote vehicle with imagery movement: A brain computer interface study," in *New Trends in Medical and Service Robotics*. Cham, Switzerland: Springer, 2019, pp. 186–195.
- [109] L. G. Hernández, O. M. Mozos, J. M. Ferrández, and J. M. Antelis, "EEG-based detection of braking intention under different car driving conditions," *Front. Neuroinformat.*, vol. 12, p. 29, May 2018.
- [110] J. Zhuang, K. Geng, and G. Yin, "Ensemble learning based brain-computer interface system for ground vehicle control," *IEEE Trans. Syst., Man, Cybern., Syst.*, early access, Dec. 9, 2019, doi: 10.1109/TSMC.2019.2955478
- [111] L. Bi, H. Wang, T. Teng, and C. Guan, "A novel method of emergency situation detection for a brain-controlled vehicle by combining EEG signals with surrounding information," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 26, no. 10, pp. 1926–1934, Oct. 2018.
- [112] T. Teng, L. Bi, and Y. Liu, "EEG-based detection of driver emergency braking intention for brain-controlled vehicles," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 6, pp. 1766–1773, Jun. 2018.
- [113] S. Haufe, M. S. Treder, M. F. Gugler, M. Sagebaum, G. Curio, and B. Blankertz, "EEG potentials predict upcoming emergency brakings during simulated driving," *J. Neural Eng.*, vol. 8, no. 5, 2011, Art. no. 056001.
- [114] I.-H. Kim, J.-W. Kim, S. Haufe, and S.-W. Lee, "Detection of braking intention in diverse situations during simulated driving based on EEG feature combination," *J. Neural Eng.*, vol. 12, no. 1, 2015, Art. no. 016001.
- [115] S. Haufe *et al.*, "Electrophysiology-based detection of emergency braking intention in real-world driving," *J. Neural Eng.*, vol. 11, no. 5, 2014, Art. no. 056011.
- [116] D. Göhring, D. Latotzky, M. Wang, and R. Rojas, "Semi-autonomous car control using brain computer interfaces," in *Intelligent Autonomous Systems 12* (Advances in Intelligent Systems and Computing), vol. 194, S. Lee, H. Cho, K. J. Yoon, and J. Lee, Eds. Berlin, Germany: Springer, 2013. [Online]. Available: https://doi.org/10.1007/978-3-642-33932-5_37
- [117] L. Bi, X.-A. Fan, K. Jie, T. Teng, H. Ding, and Y. Liu, "Using a headup display-based steady-state visually evoked potential brain–computer interface to control a simulated vehicle," *IEEE Trans. Intell. Transp. Syst.*, vol. 15, no. 3, pp. 959–966, Jun. 2014.
- [118] L. Bi, X.-A. Fan, N. Luo, K. Jie, Y. Li, and Y. Liu, "A head-up displaybased P300 brain-computer interface for destination selection," *IEEE Trans. Intell. Transp. Syst.*, vol. 14, no. 4, pp. 1996–2001, Dec. 2013.
- [119] X.-A. Fan, L. Bi, T. Teng, H. Ding, and Y. Liu, "A brain-computer interface-based vehicle destination selection system using P300 and SSVEP signals," *IEEE Trans. Intell. Transp. Syst.*, vol. 16, no. 1, pp. 274–283, Feb. 2015.
- [120] L. Bi, Y. Lu, X. Fan, J. Lian, and Y. Liu, "Queuing network modeling of driver EEG signals-based steering control," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 25, no. 8, pp. 1117–1124, Aug. 2017.
- [121] Y. Lu, L. Bi, and H. Li, "Model predictive-based shared control for brain-controlled driving," *IEEE Trans. Intell. Transp. Syst.*, vol. 21, no. 2, pp. 630–640, Feb. 2020.
- [122] A. Hekmatmanesh, H. Wu, F. Jamaloo, M. Li, and H. Handroos, "A combination of CSP-based method with soft margin SVM classifier and generalized RBF kernel for imagery-based brain computer interface applications," *Multimedia Tools Appl.*, vol. 79, pp. 17521–17549, Feb. 2020.

- [123] A. Hekmatmanesh, H. Wu, A. Motie-Nasrabadi, M. Li, and H. Handroos, "Combination of discrete wavelet packet transform with detrended fluctuation analysis using customized mother wavelet with the aim of an imagery-motor control interface for an exoskeleton," *Multimedia Tools Appl.*, vol. 78, pp. 30503–30522, May 2019.
- [124] A. Hekmatmanesh, R. M. Asl, H. Handroos, and H. Wu, "Optimizing largest Lyapunov exponent utilizing an intelligent water drop algorithm: A brain computer interface study," in *Proc. 5th Int. Conf. Event Based Control Commun. Signal Process. (EBCCSP)*, Vienna, Austria, 2019, pp. 1–5.
- [125] A. Hekmatmanesh, R. M. Asl, H. Wu, and H. Handroos, "EEG control of a bionic hand with imagination based on chaotic approximation of largest Lyapunov exponent: A single trial BCI application study," *IEEE Access*, vol. 7, pp. 105041–105053, 2019.
- [126] A. Hekmatmanesh, F. Jamaloo, H. Wu, H. Handroos, and A. Kilpeläinen, "Common spatial pattern combined with kernel linear discriminate and generalized radial basis function for motor imagery-based brain computer interface applications," in *Proc. AIP Conf.*, vol. 1956, 2018, Art. no. 020003. [Online]. Available: https://doi.org/10.1063/1.5034255
- [127] A. Hekmatmanesh, "Investigation of EEG signal processing for rehabilitation robot control," Ph.D. dissertation, Dept. Brain-Comput. Interfaces, Lappeenranta-Lahti Univ. Technol., Lappeenranta, Finland, 2019.
- [128] A. Hekmatmanesh *et al.*, "Sleep spindle detection and prediction using a mixture of time series and chaotic features," *Adv. Elect. Electron. Eng.*, vol. 15, no. 3, pp. 435–447, 2017.
- [129] A. Zafar, U. Ghafoor, M. J. Khan, and K.-S. Hong, "Drone control using functional near-infrared spectroscopy," in *Proc. 15th Int. Conf. Elect. Eng. Electron. Comput. Telecommun. Inf. Technol. (ECTI-CON)*, Chiang Rai, Thailand, 2018, pp. 384–387.
- [130] M. J. Khan and K.-S. Hong, "Hybrid EEG-fNIRS-based eightcommand decoding for BCI: Application to quadcopter control," *Front. Neurorobot.*, vol. 11, p. 6, Feb. 2017.
- [131] T. Shi, H. Wang, and C. Zhang, "Brain computer interface system based on indoor semi-autonomous navigation and motor imagery for unmanned aerial vehicle control," *Expert Syst. Appl.*, vol. 42, no. 9, pp. 4196–4206, 2015.
- [132] J.-H. Jeong, K.-H. Shim, D.-J. Kim, and S.-W. Lee, "Brain-controlled robotic arm system based on multi-directional CNN-BiLSTM network using EEG signals," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 28, no. 5, pp. 1226–1238, May 2020.
- [133] K. Volkova, M. A. Lebedev, A. Kaplan, and A. Ossadtchi, "Decoding movement from electrocorticographic activity: A review," *Front. Neuroinformat.*, vol. 13, p. 74, Dec. 2019. [Online]. Available: https://www.frontiersin.org/article/10.3389/fninf.2019.00074
- [134] S. Gopal, S. I. Ryu, B. M. Yu, A. Afshar, and K. V. Shenoy, "A highperformance brain-computer interface," *Nature*, vol. 442, pp. 195–198, Jul. 2006.
- [135] S. I. Ryu and K. V. Shenoy, "Human cortical prostheses: Lost in translation?" *Neurosurg. Focus*, vol. 27, no. 1, p. E5, Jul. 2009. [Online]. Available: https://thejns.org/focus/view/journals/neurosurgfocus/27/1/article-pE5.xml
- [136] S. Burwell, M. Sample, and E. Racine, "Ethical aspects of brain computer interfaces: A scoping review," *BMC Med. Ethics*, vol. 60, no. 18, p. 60, 2017.
- [137] A. Maxmen, "Self-driving car dilemmas reveal that moral choices are not universal," *Nature*, vol. 562, pp. 469–470, Oct. 2018.
- [138] L. Drew, "The ethics of brain-computer interfaces," Nature, vol. 571, pp. S19–S21, Jul. 2019.
- [139] R. Yuste *et al.*, "Four ethical priorities for neurotechnologies and AI," *Nature*, vol. 551, no. 7679, pp. 159–163, Nov. 2017. [Online]. Available: https://doi.org/10.1038/551159a
- [140] P. R. Roelfsema, D. Denys, and P. C. Klink, "Mind reading and writing: The future of neurotechnology," *Trends Cogn. Sci.*, vol. 22, no. 7, pp. 598–610, 2018. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1364661318300925
- [141] A. Jobin, M. Ienca, and E. Vayena, "The global landscape of AI ethics guidelines," *Nat. Mach. Intell.*, vol. 1, pp. 389–399, Sep. 2019.
- [142] D. Fang, Y. Qian, and R. Q. Hu, "Security for 5G mobile wireless networks," *IEEE Access*, vol. 6, pp. 4850–4874, 2018.
- [143] R. Khan, P. Kumar, D. N. K. Jayakody, and M. Liyanage, "A survey on security and privacy of 5G technologies: Potential solutions, recent advancements, and future directions," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 1, pp. 196–248, 1st Quart., 2020.

- [144] M. Wang, T. Zhu, T. Zhang, J. Zhang, S. Yu, and W. Zhou, "Security and privacy in 6G networks: New areas and new challenges," *Digit. Commun. Netw.*, vol. 6, no. 3, pp. 281–291, 2020. [Online]. Available: http://www.sciencedirect.com/science/article/pii/S2352864820302431
- [145] S. L. Bernal, A. H. Celdrán, L. F. Maimó, M. T. Barros, S. Balasubramaniam, and G. M. Pérez, "Cyberattacks on miniature brain implants to disrupt spontaneous neural signaling," *IEEE Access*, vol. 8, pp. 152204–152222, 2020.
- [146] S. A. Hamad, Q. Z. Sheng, W. E. Zhang, and S. Nepal, "Realizing an Internet of secure things: A survey on issues and enabling technologies," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 2, pp. 1372–1391, 2nd Quart., 2020.
- [147] M. U. Hassan, M. H. Rehmani, and J. Chen, "Differential privacy techniques for cyber physical systems: A survey," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 1, pp. 746–789, 1st Quart., 2020.
- [148] I. Makhdoom, M. Abolhasan, J. Lipman, R. P. Liu, and W. Ni, "Anatomy of threats to the Internet of Things," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 2, pp. 1636–1675, 2nd Quart., 2019.
- [149] M. Sookhak, H. Tang, Y. He, and F. R. Yu, "Security and privacy of smart cities: A survey, research issues and challenges," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 2, pp. 1718–1743, 2nd Quart., 2019.



Michael Taynnan Barros (Member, IEEE) received the Ph.D. degree in computer science from the Waterford Institute of Technology in 2016. He has been an Assistant Professor (Lecturer) with the School of Computer Science and Electronic Engineering, University of Essex, U.K., since June 2020. He is also a MSCA-IF Research Fellow (parttime) with the Tampere University, Finland. He has over 70 research peer-reviewed scientific publications in top journals and conferences, such as *Nature Scientific Reports*, IEEE TRANSACTIONS ON

COMMUNICATIONS, IEEE TRANSACTIONS ON VEHICULAR TECHNOLOGY, in the areas of molecular and unconventional communications, biomedical engineering, bionano science and beyond 5G. He received the CONNECT Prof. Tom Brazil Excellence in Research Award in 2020. He is awarded the European Commission's Marie Skłodowska-Curie Individual Fellowship from 2019 to 2021, the Irish Research Council's Government of Ireland Postdoctoral Fellow from 2016 to 2018, and the Enterprise Ireland's Commercialization Funding from 2018 to 2019. Since 2020, he has been a Review Editor for the *Frontiers in Communications* and *Networks* in the area of unconventional communications. He also served as a Guest Editor for the IEEE TRANSACTIONS ON MOLECULAR, BIOLOGICAL AND MULTI-SCALE COMMUNICATIONS and *Digital Communications Networks*.



Renan Cipriano Moioli received the B.Sc. and M.Sc. degrees in electrical engineering from the University of Campinas, Brazil, in 2006 and 2008, respectively, and the Doctoral degree in cognitive science from the University of Sussex, U.K., in 2013. He is currently Assistant Professor of Computational Intelligence with the Digital Metropolis Institute, Federal University of Rio Grande do Norte and a Member of the Multidisciplinary Bioinformatics Environment (BioME). His projects develop and explore computational intelligence tools in the con-

text of neuroengineering, from the construction of brain-machine interfaces to the search of neural patterns in cognitive processes, with a focus on health applications and broad industry sectors. He has experience in computational intelligence, machine learning, signal processing, and autonomous robotics. In 2018, he received the Royal Society Newton Advanced Fellowship. He is also a member of IEEE Bioinformatics and Bioengineering Technical Committee.



Pedro H. J. Nardelli (Senior Member, IEEE) received the B.S. and M.Sc. degrees in electrical engineering from the State University of Campinas, Brazil, in 2006 and 2008, respectively, and the Doctoral degree from University of Oulu, Finland, and State University of Campinas in 2013, following a dual degree agreement. He is currently Associate Professor (tenure track) of IoT in Energy Systems with LUT University, Finland, and holds a position of Academy of Finland Research Fellow with a project called Building the Energy Internet as a

large-scale IoT-based cyber-physical system that manages the energy inventory of distribution grids as discretized packets via machine-type communications (EnergyNet). He leads the Cyber-Physical Systems Group, LUT, and is a Project Coordinator of the CHIST-ERA European consortium Framework for the Identification of Rare Events via Machine Learning and IoT Networks (FIREMAN) and of the project Swarming Technology for Reliable and Energy-aware Aerial Missions (STREAM) supported by Jane and Aatos Erkko Foundation. He is also Docent with the University of Oulu in the topic of "Communications Strategies and Information Processing in Energy Systems." His research focuses on wireless communications particularly applied in industrial automation and energy systems.



Walid Saad (Fellow, IEEE) received the Ph.D. degree from the University of Oslo in 2010. He is currently a Professor with the Department of Electrical and Computer Engineering, Virginia Tech, where he leads the Network Science, Wireless, and Security Laboratory. His research interests include wireless networks, machine learning, game theory, security, unmanned aerial vehicles, cyber-physical systems, and network science. He is also the recipient of the NSF CAREER Award in 2013, the AFOSR Summer Faculty Fellowship in 2014, and

the Young Investigator Award from the Office of Naval Research in 2015. He was the author/coauthor of ten conference best paper awards at WiOpt in 2009, ICIMP in 2010, IEEE WCNC in 2012, IEEE PIMRC in 2015, IEEE SmartGridComm in 2015, EuCNC in 2017, IEEE GLOBECOM in 2018, IFIP NTMS in 2019, IEEE ICC in 2020, and IEEE GLOBECOM in 2020. He is the recipient of the 2015 Fred W. Ellersick Prize from the IEEE Communications Society, the 2017 IEEE ComSoc Best Young Professional in Academia Award, the 2018 IEEE ComSoc Radio Communications Committee Early Achievement Award, and the 2019 IEEE ComSoc Communication Theory Technical Committee. He was also a coauthor of the 2019 IEEE Communications Society Young Author Best Paper and the 2021 IEEE Communications Society Young Author Best Paper. From 2015 to 2017, he was named the Stephen O. Lane Junior Faculty Fellow at Virginia Tech and he was named College of Engineering Faculty Fellow in 2017. He received the Dean's Award for Research Excellence from Virginia Tech in 2019. He currently serves as an Editor for the IEEE TRANSACTIONS ON MOBILE COMPUTING, the IEEE TRANSACTIONS ON NETWORK SCIENCE AND ENGINEERING, and the IEEE TRANSACTIONS ON COGNITIVE COMMUNICATIONS AND NETWORKING. He is an Editor-at-Large for the IEEE TRANSACTIONS ON COMMUNICATIONS. He is an IEEE Distinguished Lecturer.



Amin Hekmatmanesh received his bachelor's degree in electrical engineering from the Science and Research Department, Fars University, Shiraz, Iran, in 2010, the master's degree in biomedical engineering from Shahed University, Tehran, Iran, in 2013, and the Ph.D. degree in brain-controlled ankle foot and hand orthosis and control of mobile vehicle robots using imaginary movements based on the EEG from the Laboratory of Intelligent Machines, Lappeenranta University of Technology in 2019. His master's thesis was about analyzing Sleep EEG

signal processing, memory consolidation, learning and negative emotional memory. Since 2020, he has been working as a Postdoctoral Researcher on heavy machine operator's health monitoring and signal processing for simulators with the Laboratory of Intelligent Machines, Lappeenranta University of Technology.



Pedro E. Gória Silva received the B.Sc. degree in telecommunication engineering and the M.Sc. degree in telecommunications from the National Telecommunications Institute, Brazil, in 2017 and 2020, respectively, where he is currently pursuing the Ph.D. degree in partnership with the School of Energy Systems, LUT University, Finland. In the business world, he has more than eight years of experience in the high-tech industrial sector. He has served in executive, managerial, and technical positions. The main areas he worked on are research

and development, engineering and new products, maintenance and installation of electronic equipment, and project management. His research currently includes general aspects of digital transmission, brain-type communication, chaos-based communication, mobile communication, and fading channels.



Arthur Sousa de Sena (Student Member, IEEE) received the B.Sc. degree in computer engineering and the M.Sc. degree in teleinformatics engineering from the Federal University of Ceará, Brazil, in 2017 and 2019, respectively. He is currently pursuing the Doctoral degree in electrical engineering with the School of Energy Systems, LUT University, Finland. From 2014 to 2015, he studied Computer Engineering as an exchange student with the Illinois Institute of Technology, USA. He is also a Researcher with the Cyber-Physical Systems

Group, LUT, where he has been actively working in the wireless communication field, having already published several papers in prestigious journals and conferences. His research interests include signal processing, mobile communications systems, non-orthogonal multiple access techniques, intelligent metasurfaces, and massive MIMO.



Merim Dzaferagic received the bachelor's and master's degrees in telecommunications engineering from the University of Sarajevo, Bosnia and Herzegovina, in 2011 and 2013, respectively, and the Ph.D. degree from Trinity College, Dublin, Ireland, in 2020, where he is currently a Postdoctoral Researcher with CONNECT. As part of his Ph.D., he has been working on the concept of emergence, distributed optimization and self-organization solutions for communication networks. His work focuses on complex system science and machine learning

applied to communication networks. His work has been focused on the impact that the topological structures have on the overall network performance, as well as the application of different machine learning approaches to support the operation of industrial IoT networks.



Harun Siljak (Senior Member, IEEE) received the bachelor's and master's degrees in electronics and control engineering from the University of Sarajevo, Bosnia and Herzegovina, in 2010 and 2012, respectively, and the Ph.D. degree in electronics and electrical engineering from the International Burch University Sarajevo, Bosnia and Herzegovina, in 2015. He is currently an Assistant Professor of Embedded Systems, Control and Optimization with Trinity College, Dublin, Ireland, where he previously worked as a Marie Curie Postdoctoral

Fellow (EDGE Cofund). He has published more than 40 peer reviewed papers in communications, computation, control, and mathematics, and more than 15 media articles on science, technology, and society. His research interests span from complex networks and cyber-physical systems to unconventional methods of computation, communication, and control. He is an Active Science Communicator and Science Fiction Author, combining his research and storytelling. He is an Associate Editor of *EURASIP Journal on Wireless Communications and Networking*, a Review Editor of *Non-Conventional Communications and Networks*, Frontiers in Communications and Networks, and the Science Communication Officer for COST Action MAT-DYN-NET and Marie Curie Alumni Association's Western Balkans Chapter.



Werner Van Leekwijck received the B.S. and M.Sc. degrees in electrical engineering from the University of Leuven in 1988 and 1990, and the M.Sc. degree in knowledge and information technology from the University of Ghent in 1994. After more than 20 years of industrial research with Alcatel and Bell Labs and leading numerous research projects in collaboration with academia. He is currently a Senior Investigator with the Internet & Data Lab (IDLab), Imec Research Group, University of Antwerp, with focus on wireless networking and artificial intelli-

gence. He leads the group on sustainable neuro-inspired computing systems, investigating low-power 3rd generation neural networks and beyond.



Dick Carrillo Melgarejo (Member, IEEE) received the B.Sc. degree (Hons.) in electronics and electrical engineering from San Marcos National University, Perú, in 2004, and the M.Sc. degree in electrical engineering from the Pontifical Catholic University of Rio de Janeiro, Brazil, in 2008. He is currently pursuing the double Ph.D. degree in electrical engineering with LUT, Finland, and UNICAMP, Brazil. From 2008 to 2018, he participated on many research and development projects based on cellular networks, such as WiMAX, LTE, and LTE-A IoT

applications. In 2019, he was a Visiting Research Fellow with the CTTC, Spain. His research interests are applications protocols enabled by mobile technologies beyond 5G, such as brain-to-brain communication and industrial protocols such as IEC-61850. In the communication spectrum, his main interest are waveforms, energy harvesting, reconfigurable intelligent surfaces, cell free massive MIMO, and grant-free access.



Steven Latré (Senior Member, IEEE) received the M.Sc. degree in computer science from Ghent University, Belgium, and the Ph.D. degree in computer science engineering from Ghent University, with the title "Autonomic Quality of Experience Management of Multimedia Services." He is currently a Professor with the University of Antwerp and the Director of the Research Centre IMEC, Belgium. He is leading the IDLab Antwerp Research Group (over 100 members), which is performing applied and fundamental research in the area of

communication networks and distributed intelligence. He has authored or coauthored more than 100 articles published in international journals or in the proceedings of international conferences. His personal research interests are in the domain of machine learning and its application to wireless network optimization. He is a recipient of the IEEE COMSOC Award for Best Ph.D. in network and service management 2012, the IEEE NOMS Young Professional Award in 2014, and is a member of the Young Academy Belgium.